

Supplemental Guidance on Human Health Risk Assessment of Indoor Settled Dust (HHRA_{DUST})



Federal Contaminated Site Risk Assessment in Canada



Health Canada is the federal department responsible for helping the people of Canada maintain and improve their health. We assess the safety of drugs and many consumer products, help improve the safety of food, and provide information to Canadians to help them make healthy decisions. We provide health services to First Nations people and to Inuit communities. We work with the provinces to ensure our health care system serves the needs of Canadians.

Également disponible en français sous le titre :

L'évaluation des risques pour les sites contaminés fédéraux au Canada : Guide supplémentaire sur l'évaluation des risques pour la santé humaine liés aux dépôts de poussière intérieure (ÉRSH_{POUSSIÈRE})

To obtain additional information, please contact:

Health Canada Address Locator 0900C2 Ottawa, ON K1A 0K9 Tel.: 613-957-2991 Toll free: 1-866-225-0709 Fax: 613-941-5366 TTY: 1-800-465-7735 Email: publications@hc-sc.gc.ca

© Her Majesty the Queen in Right of Canada, as represented by the Minister of Health, 2018

Publication date: February 2018

This publication may be reproduced for personal or internal use only without permission provided the source is fully acknowledged.

Cat.: H144-50/2018E-PDF ISBN: 978-0-660-23679-7 Pub.: 170262

FEDERAL CONTAMINATED SITE RISK ASSESSMENT IN CANADA

SUPPLEMENTAL GUIDANCE ON HUMAN HEALTH RISK ASSESSMENT OF INDOOR SETTLED DUST (HHRA_{DUST})

February 2018

Prepared by: Contaminated Sites Division Safe Environments Directorate

TABLE OF CONTENTS

PRE	FACE.		iv		
ABBF	REVIA	TIONS AND ACRONYMS	V		
1.0 INTRODUCTION					
	1.1	Purpose	1		
	1.2	Background	2		
	1.3	Scope of this Document	3		
	1.4	Layout of Document	3		
2.0	DEVI	ELOPING A CONCEPTUAL MODEL FOR ASSESSMENT OF INDOOR DUST	4		
	2.1	When to Consider Assessment of Indoor Dust	4		
	2.2	Assessment of COPCs in Indoor Dust	5		
3.0	SAM	PLING OF INDOOR DUST	5		
	3.1	Characterization of Indoor Dust	6		
		3.1.1 Determination of Percentage of Dust From Soils	7		
	3.2	Indoor Dust Sampling Method	8		
		3.2.1 Wipe Sampling (Surface Area Measurement)	.10		
		3.2.2 Vacuum Sampling (Bulk Dust Measurement)	.10		
		3.2.3 Wipe Versus Vacuum Sampling	. 11		
		3.2.4 Recommendations for Dust Sampling	.12		
	3.3	Laboratory Analysis	.13		
	3.4	Quality Assurance and Quality Control	.13		
4.0	INCL	USION OF COPCS IN INDOOR DUST IN AN HHRA	.14		
	4.1	Selecting COPCs in Dust	.15		
		4.1.1 Screening Against DSC _{HH} —Surface Area and Bulk Dust	.16		
		4.1.2 Screening Against Baseline Dust Concentrations	.16		
		4.1.3 Selection of COPCs	.19		
	4.2	Receptor Characterization	.19		
	4.3	Exposure Characterization Assumptions	.22		

4.4		Concentrations of COPCs in Indoor Dust—Use of Appropriate Statistics	
		4.4.1 Use of Maximum Concentration Versus Mean Concentration	24
		4.4.2 Incorporation of Data with Non-Detected Concentrations	24
		4.4.3 Data Requirements	24
	4.5	Toxicity Assessment	25
	4.6	Bioavailability of COPCs in Indoor Dust	25
	4.7	Risk Characterization	25
5.0	CON	CLUSIONS	26
6.0	REF	ERENCES	27
APPE	ENDIX	1	34
APPE	ENDIX	. II	39

LIST OF TABLES

TABLE 1: Examples of Published Sampling Protocols for Indoor Dust	9
TABLE 2: Possible Advantages and Disadvantages of Vacuum Versus Surface Wipe Sampling Methods*	12
TABLE 3: Examples of Published Canadian Baseline Dust Data	17
TABLE 4: Example Receptor Characteristic Parameters to Derive Dust Screening Concentrations*	20
TABLE 5: Example Exposure Parameters to Derive Dust Screening Concentrations*	22
TABLE 6: Site-Specific Input Parameters for HHRAs	24

PREFACE

The Federal Contaminated Sites Action Plan (FCSAP) is a program of the Government of Canada designed to ensure improved and continuing federal environmental stewardship as it relates to contaminated sites located on federally owned or operated properties or non-federal lands for which the federal government has accepted full responsibility. Guidance documents on human health risk assessment (HHRA) prepared by the Contaminated Sites Division of Health Canada, may be obtained by contacting the Contaminated Sites Division at: <u>cs-sc@hc-sc.gc.ca</u>.

This guidance document was prepared for the benefit of custodial departments who are assessing indoor settled dust as part of a risk assessment. As is common with any national guidance, this document may not satisfy all custodian site specific requirements. As the practice of HHRA advances and as the FCSAP proceeds, new and updated information on various aspects of HHRA will be published. Therefore, revisions to this document will likely be necessary from time to time to reflect new information. Health Canada should be consulted at the address below to confirm that the version of the document in your possession is the most recent edition and that the most recent assumptions, parameters, etc., are being used.

In addition, Health Canada requests that feedback on this document be directed to the Contaminated Sites Division, Safe Environments Directorate, Healthy Environments and Consumer Product Safety, Health Canada, e-mail: <u>cs-sc@hc-sc.gc.ca</u>.

See also: www.canada.ca/en/health-canada/services/environmental-workplace-health/contaminated-sites.html

Health Canada acknowledges the feedback and review comments received from a variety of government, academia and industry parties during the preparation of this guidance document. Health Canada also acknowledges the extensive work completed by Meridian Environmental Inc., Millenium and Wilson Scientific under contract to Health Canada.

ABBREVIATIONS AND ACRONYMS

AAS	Atomic absorption spectrometry
ASTM	American Society for Testing and Materials
ATSDR	Agency for Toxic Substances and Disease Registry
CCME	Canadian Council of Ministers of the Environment
CHDS	Canadian House Dust Study
COPC	Contaminant of potential concern
DQO	Data quality objective
DQRA	Detailed quantitative risk assessment
DSC	Dust screening concentration
DSC _{HH}	Human health-based dust screening concentration/s
FCSAP	Federal Contaminated Sites Action Plan
GC-MS	Gas chromatography-mass spectrometry
HC	Health Canada
HEPA	High efficiency particulate air
HHRA	Human health risk assessment
HQ	Hazard quotient
HVS3	High Volume Small Surface Sampler
ICP-MS	Inductively coupled plasma-mass spectrometry
ICP-OES	Inductively coupled plasma-optical emission spectrometry
IEUBK	Integrated Exposure Uptake Biokinetic model
ILCR	Incremental lifetime cancer risk
INAA	Instrumental neutron activation analysis
MAD	Median absolute deviation
PAH	Polycyclic aromatic hydrocarbons
PBDE	Polybrominated diphenyl ether
PCB	Polychlorinated biphenyl
PQRA	Preliminary quantitative risk assessment
QA	Quality assurance
QC	Quality control
SF	Slope factor for carcinogenic potency
TDI	Tolerable daily intake
TRV	Toxicological reference value
US EPA	United States Environmental Protection Agency
US HUD	United States Department of Housing and Urban Development
VDI	Verein Deutscher Ingenieure (Association of German Engineers)
WTCWG	World Trade Center Indoor Air Task Force Working Group

1.0 INTRODUCTION

Human health risk assessment (HHRA) is an approach used by health agencies and risk assessment professionals to estimate the potential effects of chemicals on human health. HHRAs form the basis of many of the guidelines and standards developed in Canada for establishing acceptable levels of chemicals in food, water, soil and air. Although Health Canada (HC) has used HHRA principles to develop various soil quality guidelines and risk-based decisions for many years, there are no standard assumptions and methods for assessment of indoor settled dust. This guidance document is intended to provide guidance on the key issues and methods used to assess potential health risks associated with chemicals from contaminated sites that may impact settled dust, in particular at federal contaminated sites in Canada.

Dust in the indoor environment is a complex matrix formed from a variety of anthropogenic, natural, indoor and outdoor sources, to which people are exposed as a whole. Assessment of human health risks associated with chemicals at contaminated sites may include evaluation of several types of environmental media such as soils, water, air, food and dust. Many of the chemicals present at contaminated sites are also found at non-contaminated sites, originating, for example, from background natural sources (such as naturally occurring minerals) or anthropogenic chemicals used in consumer products. Interpretation of levels of chemicals in dust may be difficult. Background/baseline dust concentrations from non-contaminated areas may provide a meaningful point of comparison to interpret what is typical and atypical for a given site and to assist in assessing the impacts of contaminated soils on indoor dust.

HHRA is a dynamic science, and therefore, acceptable practices change over time with new scientific developments. Discussion with HC staff is encouraged to ensure accordance with current methods and procedures. In addition, it is stressed that HHRA should be completed only by qualified and experienced risk assessment practitioners.

1.1 Purpose

The purpose of this guidance document is to recommend standard methods, exposure parameters and equations that can be used to quantitatively assess the potential exposures and risks to people from substances that may be elevated in indoor settled dust at federal contaminated sites due to outdoor sources. Further, information is provided for the risk assessor to derive human health-based dust screening concentrations (DSC_{HH}) and to assess exposure to contaminants of potential concern (COPCs) in indoor dust in quantitative human health risk assessments at federal contaminated sites.

The approach is specifically designed for the assessment of sites that are to remain the responsibility of federal agencies and of properties for which greater consistency in risk assessment methods and interpretation of results is required. For properties being divested to a private party or to provincial or municipal government agencies, HHRAs may have to be completed in accordance with provincial/territorial regulatory requirements. Local regulatory requirements may differ from the standardized methods described herein.

This guidance document was developed for use by federal custodial departments receiving funding under the FCSAP and to supplement Health Canada's Federal Contaminated Site Risk Assessment in Canada, Part V: Guidance on Human Health Detailed Quantitative Risk Assessment for Chemicals (DQRA guidance) (HC, 2010a) and is not intended for broader application outside of the FCSAP. HHRAs at contaminated sites consider the impact that historical activities may have had on various media, including soils, groundwater, surface water and air. This guidance document was prepared to address human health risks associated with chemicals present in indoor settled dust at federal contaminated sites when this medium is deemed to be of potential concern.

Although the guidance offered here is detailed in nature, it is not designed or intended as a substitute for the sound professional judgement of a qualified and experienced risk assessment practitioner. It is recognized that many HHRAs will present unique situations not specifically addressed here. Risk assessors are encouraged to ensure that their assessments address all relevant potential risks. The methods delineated should not be viewed as a "black box" of equations and assumptions that negate the need for sound professional judgement. However, where possible and appropriate, this guidance will help provide consistency in the approach used at federal sites. Where alternative or unique approaches have been used, these should be sufficiently documented and described to enable peer review and should be evaluated for their impact on risk estimates relative to the application of the standard methods prescribed.

1.2 Background

The United States Environmental Protection Agency (US EPA, 1997) defines house dust as "a complex mixture of biologicallyderived material (animal dander, fungal spores, etc.), particulate matter deposited from the indoor aerosol, and soil particles brought in by foot traffic." The German Verein Deutscher Ingenieure (VDI, 2001) Protocol 4300 states that "there is currently no generally binding definition of the term settled house dust. To delimit the term from suspended particulate matter, it is intended to mean all types of particles which are encountered indoors in deposited form. The dust may be solids of the most varied inorganic or organic materials, which can be of natural or synthetic origin. The term includes not only fractions which originate indoors themselves, but also those which are introduced from the outside." These definitions of house dust, which highlight the deposited nature of this complex mixture of variable source, are adopted here. In this guidance, the term "house dust" is defined as above but can alternatively be referred to as "indoor settled dust," "settled dust," "indoor dust" and "dust." The term "soil" refers to outdoor soils.

Mounting evidence shows that exposure to environmental chemicals in indoor dust can be significant. An increasing number of studies have attempted to quantify chemical exposures from indoor dust, with results for certain chemicals such as lead and polybrominated diphenyl ethers (PBDEs) indicating that dust could represent a significant source of exposure for young children (Dixon *et al.*, 2009; Jones-Otazo *et al.*, 2005; Roberts *et al.*, 2009; US Department of Housing and Human Development [HUD], 2006). Correlations between blood lead levels (a measure of body burden) in children or adults and indoor dust concentrations or loadings (Dixon *et al.*, 2009; von Lindern *et al.*, 2003; Lanphear *et al.*, 1998) add weight to the importance of indoor dust as a potential source of chemical exposure. Dust may be a major source of exposure to pesticides, polycyclic aromatic hydrocarbons (PAHs), phthalates, arsenic, cadmium and chromium (Freeman *et al.*, 1997; Roberts *et al.*, 2009).

On average, Canadians spend approximately 90% of their time indoors (Matz *et al.*, 2014), which may result in greater exposure to indoor dust than to outdoor soil. The majority of time spent indoors is spent at home and other indoor locations including, but not limited to, work/school, grocery stores, shopping malls, and restaurants/bars (Matz *et al.*, 2015). People may be exposed to indoor dust via oral exposure through ingestion of settled dust from hand-to-mouth and object-to-mouth contact; via inhalation of re-suspended dust particles in indoor air; and also via dermal contact with settled dust. Elevated levels of chemicals in dust can present a potential exposure pathway at contaminated sites; this may require investigation at some sites, whereas at others the pathway may not be significant and/or operable.

In general, indoor dust consists of finer particles than does outdoor soil and contains higher organic carbon content and chemical concentrations. Finer particles adhere to the skin more effectively than coarse particles (Kissel *et al.*, 1996), resulting in higher exposures via hand-to-mouth and dermal contact. Further, characteristics of the indoor environment ensure that fewer degradation, aging and dispersive processes act on indoor dust as compared with outdoor soils, and certain indoor soft surfaces "trap" dust and result in continued exposure (Paustenbach *et al.*, 1997).

Exposures of young children to indoor settled dust are typically of greater concern given their vulnerabilities, and their exposure and behavioural characteristics. Young children frequently mouth hands, toys and other non-food items, and they crawl on floors in exploratory behaviours that may result in more skin contact, ingestion and inhalation of dust (US EPA, 2008a; Roberts *et al.*, 2009). Young children may also be more vulnerable to some chemicals in dust, given their immature systems (which may be less equipped to metabolize, detoxify and excrete chemicals), their rapid physical development and their vulnerability to developmental toxicants (US EPA, 2008a; Roberts *et al.*, 2009; Morgan *et al.*, 2005).

While the relationship between indoor air and settled dust is fluid and complex, the two are typically treated as discrete environmental media and assessed separately in HHRAs. Particles between about 1 nm and 100 µm may become suspended in air, depending on their physical properties and the forces and processes to which they are subjected (Morawska and Salthammer, 2004). Particles suspended in air may become deposited indoors as a result of particle diffusion (smaller particles) and gravitational sedimentation (larger particles), but particle deposition is highly variable, site-specific and difficult to guantify or predict (Morawska and Salthammer, 2004). Particles deposited as settled dust may also become re-suspended into indoor air. with larger particles being more readily re-entrained than smaller ones. Normal indoor activities, such as walking, moving around, playing or cleaning, have been shown to result in particle re-suspension (Morawska and Salthammer, 2004). Although some episodic inhalation of re-suspended dust (and associated chemicals) can be expected during such activities, these exposures are more appropriately assessed through an indoor air quality assessment, rather than through a settled dust assessment (with corresponding use of complex and uncertain modelling of deposition and re-suspension processes). Different air quality monitoring techniques are available that would enable assessment of temporal variations in air guality and exposure, and the appropriate type and duration of air sampling should be determined according to the needs of the air quality evaluation. Health Canada (2017) has a separate document which provides guidance on assessment of air guality. For the above reasons, this guidance deals with exposure via ingestion and dermal pathways, and excludes the inhalation pathway. It is recognized that in order to effectively assess the health risk of chemicals from contaminated sites for which indoor settled dust is an exposure source, all routes of exposure must be considered.

1.3 Scope of this Document

Residential and commercial environments were selected as the focus of this guidance document because indoor dust in these environments may be relevant to federal contaminated sites. Exposure assumptions as defined by the Canadian Council of Ministers of the Environment (CCME, 2006) and HC (2012a) for residential and commercial environments were assumed. Adult and toddler receptors are assumed to reside in residential environments, and adult employee and toddler receptors are assumed to be present in commercial environments. Other age groups may also be considered in risk assessments; however, the toddler receptor is typically the most sensitive, on the basis of ingestion rates and body weight. The "constrained use commercial" environment is assumed to be limited to adult employees. Specific exposure assumptions for these environments are available in HC (2012a). The HHRA should consider human uses of the site and apply the most appropriate land use designation.

Contaminated sites can include active working environments. Industrial or commercial environments where chemicals are used within the active work environment are typically considered under occupational worker health and safety legislation and/or regulations, and are not considered within this document.

1.4 Layout of Document

This report is organized as follows:

- Section 2 provides general guidance on when to consider assessment of exposure to chemicals in settled dust in an HHRA;
- Section 3 provides considerations for dust sampling;
- Section 4 provides a description of how to include dust as an exposure medium in an HHRA;
- Section 5 provides conclusions;
- Appendix I provides equations for calculating DSC_{HH}; and
- Appendix II provides equations for incorporating dust exposure assessment into HHRAs.

2.0 DEVELOPING A CONCEPTUAL MODEL FOR ASSESSMENT OF INDOOR DUST

A risk assessment of any contaminated site includes development of a conceptual site model at the problem formulation stage. This conceptual model will include identification of which exposure pathways, such as the indoor dust pathways, are operable at a particular site. A narrative should accompany the conceptual model to identify whether people may be exposed to chemicals in various media and the mechanisms of exposure.

Indoor dust may be considered a potential medium of interest at sites with contaminated surface soils. Concentrations of COPCs in indoor settled dust may become elevated through tracking of soil indoors, windblown dust and emissions during remediation or construction. Potential sources of indoor dust contamination should be identified in the conceptual model.

General information on problem formulation is not detailed further in this guidance document, and the reader is referred to HC's DQRA guidance (HC, 2010a) for further information, including the following:

- Screening and identification of COPCs;
- Identification and description of potential receptors;
- Identification of operable and inoperable exposure pathways;
- Description of the COPCs, critical receptor(s) and exposure pathways;
- Listing of all major assumptions used in the problem formulation; and
- Presentation of the problem formulation checklist (HC, 2012a).

2.1 When to Consider Assessment of Indoor Dust

The indoor dust ingestion and dermal exposure pathways are only considered operable if there is a source of contamination, a human receptor, a mechanism (pathway) for the contamination from the site to impact indoor dust, and subsequent exposure for people via ingestion and dermal exposure to dust. Airborne dust is assessed separately. Chemicals that are present in settled dust from sources other than the contaminated site would not be included in a contaminated site risk assessment, unless background exposure is assessed.

To assess when it is appropriate to pursue the evaluation of the indoor dust pathway, a series of questions need to be answered during the development of the conceptual model:

- 1. Does the contaminated site represent a significant source of contamination to indoor dust?
- 2. Is there a transport mechanism through which COPCs from the contaminated soil, outdoor dust and/or outdoor air at the site could impact indoor dust?
- 3. What are the exposure patterns and characteristics of the receptors with respect to indoor dust?
- 4. Do chemical concentrations in indoor dust exceed dust screening concentrations (DSCs)?

It is the responsibility of the federal custodial department with its qualified risk assessment practitioner to determine whether or not indoor dust should be sampled and included in an HHRA. Health Canada may be consulted during this process.

The assessment of indoor dust in an HHRA could be considered at federal contaminated sites under the following conditions:

- Soil track-in or airborne emissions of contaminated soils where substances exceed human health-based soil quality guidelines and which could impact settled dust;
- Comprehensive assessment of multiple environmental media is desired, and dust impacts are reasonably anticipated from contaminated sites; and,
- Local stakeholders have expressed concern over chemical concentrations inside their homes and buildings, and dust impacts can be reasonably anticipated to be from contaminated soils at or near the site.

2.2 Assessment of COPCs in Indoor Dust

COPCs are typically identified as chemicals that pose or have the potential to pose risks to human health. These chemicals are carried forward to the subsequent stages of the risk assessment (see HC [2010a] for more information). COPC concentrations in indoor dust may be measured directly or may be modelled from measured concentrations in soil.

At a contaminated site, the identification of COPCs is based on current and historical use of the property and the measurement of chemicals in environmental media, which will typically begin with an assessment of impacts to soils in environmental site investigations, as described in HC (2010a). For an assessment of indoor dust at a contaminated site, the COPCs associated with the site are identified from environmental site assessments and are discussed in the problem formulation stage of the HHRA.

Measured or modelled (predicted) concentrations of the COPCs in indoor dust can be compared with DSCs that can be calculated using the method presented in this report. If the measured or modelled concentration exceeds the DSC, the COPC would be retained for further evaluation within the HHRA.

3.0 SAMPLING OF INDOOR DUST

Before indoor dust is sampled, detection limits should be identified on a chemical-specific basis to ensure that they are adequate to support a risk assessment. The detection limits should be less than the calculated DSCs, and the risk assessor should identify the target level (e.g., the DSC or detection limit) prior to sampling and laboratory analysis.

This guidance document does not provide detailed information on modelling the impacts of soil on settled indoor dust, nor does it provide detailed information on sampling and analysis of chemicals in settled dust. The risk assessor should provide a summary of the scientifically defensible approach used to determine concentrations of chemicals in indoor dust.

The sampling plan should identify the areas of the building that will be sampled, in advance, along with supporting rationale. Areas of high exposure for toddlers or children, such as floors and window sills at hand level in frequently occupied rooms, are preferentially sampled in order to evaluate typical exposure and subsequent health risks.

Some sampling plans may also include undisturbed or infrequently cleaned areas, such as the tops of vents or ducts and behind or beneath large pieces of furniture, in order to obtain a measure of total levels of substances in settled dust in a building for determination of remediation requirements; however, these undisturbed areas are unlikely to be a significant source of exposure for people on a daily basis. If data from such areas are obtained, they should be used appropriately (i.e., given consideration of actual exposure) and discussed in the report.

Sampling considerations for indoor settled dust are discussed in the section below. As noted above, the information presented in this document is not prescriptive, and alternative methods may be used. The methods applied at a site should be summarized in the HHRA report, with potential uncertainties and limitations identified.

For the purpose of this guidance document, it is not possible to provide specific guidance on how to identify dust that is more likely to be impacted by contaminated soil; however, it is possible that dust in entryways would be impacted to a greater degree (e.g., as a result of contaminated soil tracked in on outdoor shoes), and these data may be used in interpretation of the COPC concentrations in other areas. Also to be considered in the collection of dust are the exposure of people in the buildings and the areas most likely to be used. Any areas expected to have higher COPC concentrations (such as those used for hobbies involving chemicals) should be identified. A clear rationale for sampling and analysis should be provided in the HHRA report as well as in the proposed sampling plan.

3.1 Characterization of Indoor Dust

The size of indoor settled dust particles, as well as their composition, is dependent on the source of the dust. The dust found in a building can originate from external sources such as contaminated soils, including tracked-in or re-suspended soil particles, but also from atmospheric deposition of particulates from industrial sources, forest fires, sea salt, heating plants and vehicular traffic (Davis and Gulson, 2005; Freeman *et al.*, 1997; Lioy *et al.* 2002; Morawska and Salthammer, 2004; Riley *et al.*, 2002). Additionally, indoor dust may originate from indoor sources, such as cigarette smoke, candles, cooking and heating residues, household appliances, maintenance products, textile fibres, human and animal skin particles and hair/fur, food crumbs, microorganisms, fungal spores, dust mite excreta, moulds, pollen, ash, soot, paper fibres and building materials (Afshari *et al.*, 2005; Agency for Toxic Substances and Disease Registry, 2002; Fan and Zhang, 2001; Klepeis *et al.*, 2003; Lazaridis *et al.*, 2008; Li *et al.*, 1993; Lioy *et al.*, 2002; Mølhave *et al.*, 2000; Morawska and Salthammer, 2004; Riley *et al.*, 2002; Rodes *et al.*, 1990; Zhang *et al.*, 2000). While airborne dust concentrations are often measured as PM_{2.5} (particles with a diameter of 2.5 µm or less) or PM₁₀ (particles with a diameter of 10 µm or less), the particle size of indoor dust can be highly variable.

Different sources of dust often have different particle size distributions, and different sizes and types of dust particles may also have different chemical characteristics, affecting the adsorption of chemicals to their surfaces. The surface area to mass ratio is greater in smaller particles, which may result in higher chemical concentrations in smaller particle size fractions (when expressed as mass of chemical per mass of dust). This is supported by a study of American residences (Lewis *et al.*, 1999), which found that concentrations of pesticides and PAH in indoor dust increased with decreasing particle size. The bioaccessibility of some metals has been shown to be influenced by particle size and also organic carbon content (Rasmussen *et al.*, 2008). In this study of vacuum dust from Ottawa, as particle size fraction decreased, bioaccessibility increased for nickel and copper but not for zinc.

Another difference with respect to indoor dust is that the organic carbon content is typically much higher than that in soil and often higher in the fine particle fraction than the coarse fraction. For example, Rasmussen (2004) found the organic carbon content to be approximately 4% in topsoil particles less than 53 µm, while the same size fraction in urban house dust from the same area had an organic carbon content of approximately 27.5%. This may contribute to higher concentrations of metals being associated with smaller particles as a result of binding with organic carbon (Rasmussen, 2004).

The intent of characterizing indoor settled dust at contaminated sites is to assess the contribution of the external contaminated media to the indoor COPC concentration in dust. It is important to note that for some substances (such as lead) that may be elevated in soils at a contaminated site, there may also be indoor sources not associated with the contaminated site, such as flaking and disintegration of interior paint. The COPCs in soils at a site (i.e., chemicals for which the maximum measured on-site concentration exceeds the applicable human health-based soil quality guideline) that are also present in indoor dust are included in a risk assessment as part of the overall assessment of exposure to the chemical at the site. In general, when collecting dust at contaminated sites for analysis, consideration needs to be given to characterizing the indoor dust most likely impacted by external media, rather than by indoor sources. This is important for risk management of contaminated sites and consideration of soil remediation options to reduce exposure.

This guidance document does not provide information regarding characterization of dust associated with indoor sources; however, the risk assessor will need to address this confounding factor as one of the uncertainties in the risk assessment.

3.1.1 Determination of Percentage of Dust From Soils

The infiltration of particulate matter from outdoors to indoors is affected by several factors, including outdoor particle concentrations, the building air exchange rate, the deposition rate of dust particles and the efficiency with which the particles can infiltrate the building. Abt *et al.* (2000) reported that 63% to 92% of indoor particles in the 0.02 μ m to 0.03 μ m range originated from outdoor sources, whereas in the 2 μ m to 10 μ m range only 20% to 43% of indoor particles were from outdoor sources. Data were included only for days when the air exchange rate was less than 1 exchange/hour to minimize the influence of air exchange; this rate is consistent with typical assumptions made for Canadian residences. Seasonal variation also affects infiltration of particles from outdoors (Long *et al.*, 2001; Morawska and He, 2004).

Some studies have shown indoor dust to have higher concentrations of metals than outdoor dust or soil (Rasmussen *et al.*, 2001; Oomen and Lijzen, 2004; Davis and Gulson, 2005). Rasmussen *et al.* (2001) reported that indoor dust differed significantly in multi-element composition from garden soil and street dust. In this study, the ratios of the geometric mean concentration of metals in indoor dust to their concentration in garden soils (i.e., the mean metal concentration observed in house dust [μ g/g]/the mean metal concentration [μ g/g] observed in garden soil) were calculated as 5.5, 16.4 and 3.4 for lead, cadmium and nickel respectively. The authors also concluded that dust generated from indoor sources may be significant for lead, cadmium, antimony and mercury. In a literature review of numerous studies and guidelines, Paustenbach *et al.* (1997) cited dust/soil concentration ratios for lead ranging from 0.3 to 9.2 in several North American and European studies, and pointed to evidence that indoor dust concentrations of cadmium, copper, zinc, pesticides and PAHs are often higher than in soil. While these observations may be indicative of additional indoor sources of these chemicals, the unique features of indoor dust as compared with soil (e.g., organic carbon content, degradation processes) may also explain higher chemical concentrations indoors (Paustenbach *et al.*, 1997).

Numerous studies have investigated the relative contribution of exterior soil to indoor dust. Oomen and Lijzen (2004) and Van Holderbeke *et al.* (2008) indicated that the contribution of exterior soil to interior house dust has been observed to range from 8% to >80%. The majority of the cited studies, however, report that 30% to 50% of house dust originates from exterior soil (Oomen and Lijzen, 2004). The US EPA (1998) Integrated Exposure Uptake Biokinetic (IEUBK) model for lead used a default value of 70% (i.e., 0.7 g soil/g dust). According to the US EPA (1998), the default value of 70% is based on an analysis of empirical data describing the relationship between soil and dust lead concentrations measured from a variety of residential communities. Oomen and Lijzen (2004) concluded that 30% to 70% of exterior soil present in house dust is likely a good approximation for evaluating exposure to indoor house dust. Cornelis and Swartjes (2007) recommended the use of 50% exterior soil in interior dust for residential scenarios with a garden or vegetable garden, an estimate also supported by analysis in Paustenbach *et al.* (1997). In homes without a garden, Cornelis and Swartjes (2007) suggest attributing 25% to exterior soil present in interior dust, although this estimate is associated with uncertainty. According to the literature, estimates of soil track-in contributions to dust vary considerably and are not recommended for predicting indoor exposures. For this reason, HC does not recommend a specific assumption for the percentage of dust that may originate from outdoor soils. Where it is important to know the site-specific ratio for risk management purposes, site-specific data should be collected.

Soil chemical concentrations and soil track-in estimates do not appear to be reliable general predictors of indoor dust concentrations, as such models have been shown to underestimate indoor exposures. Specifically, pilot studies for the Canadian House Dust Study (CHDS) demonstrated that predictive models (e.g., the IEUBK model mentioned earlier) do not hold where there is a lack of spatial correlation between elemental concentrations in indoor dust and outdoor soil, indicating a high degree of variability of indoor/outdoor concentration ratios (Rasmussen *et al.*, 2001; Rasmussen, 2004). In the city of Ottawa, such models underestimated the risk of indoor lead exposure by a factor of 8 and indoor nickel exposure by a factor of 9. When metal bioaccessibility was considered, the errors in the predictive models were even greater. Chemical concentrations in indoor dust are therefore not readily predicted using models based on either soil or outdoor particulate concentrations.

3.2 Indoor Dust Sampling Method

There are two primary methods of measuring indoor settled dust concentrations: surface wipe sampling (surface area measurement) and vacuum sampling (bulk dust measurement). Other methods, such as monitoring of heating, ventilation and air conditioning or other air ventilation systems (US EPA, 2005) or sampling attic dust (Lioy *et al.* 2002), can also be applied but are less commonly used and are not discussed further in this document.

Analytical considerations should be identified prior to sampling, on a chemical-specific basis, to ensure that the appropriate amount of dust sample is collected from each area of interest. Consultation with the analytical laboratory may be required before selecting the sampling technique and initiating sampling.

Wipe and vacuum sampling are often both used, sometimes in conjunction with air monitoring, in order to best utilize the strengths of each method and determine the distribution of chemicals. Field procedures generally involve isolating a known surface area to be sampled, either by taping off a square or placing down a template; obtaining the sample; and then transporting the sample to a laboratory for analysis.

Both wipe and vacuum sampling methods can provide surface loading levels (mass of chemical per unit area). Vacuum samples can also be analyzed to obtain the concentration of chemical by weight (e.g., $\mu g/g$), which allows for direct comparison with concentrations of COPCs in soil. Additionally, bulk samples from vacuum sampling can be used to assess particle size fractions. Some wipe sampling methodologies may also allow for determination of concentration by weight (Lioy *et al.*, 2002).

Several sampling protocols exist for procuring indoor settled dust samples by either wipe or vacuum methods. Health Canada does not specifically recommend any sampling protocols. The report should include the rationale for selecting a sampling protocol, identifying how the data will be used to assess exposure and how representative the samples are expected to be with regard to the risk assessment. Some protocols are listed in Table 1 as examples. Other sampling methods may be used and should be appropriately referenced.

Table 1: Exam	ples of Published	Sampling	Protocols for	or Indoor Dust
		• a mp mg		

Government Agency	Report Name	Report Date	Comments
American Society for Testing and Materials (ASTM)	ASTM E1728–10 Standard Practice for Collection of Settled Dust Samples Using Wipe Sampling Methods for Subsequent Lead Determination	2010	ASTM standard for dust wipe sampling for subsequent determination of lead.
ASTM	ASTM E1792–03 Standard Specification for Wipe Sampling Materials for Lead in Surface Dust	2011	ASTM specification for wipes that are used to collect settled dust on surfaces for the subsequent determination of lead.
ASTM	ASTM D5438–11 Standard Practice for Collection of Floor Dust for Chemical Analysis	2011	ASTM method for sampling dust from bare floors and carpet for chemical analyses.
California Department of Health Services	Exposure Investigation Protocol Dust Sampling in Building 240	April 2006 (revised)	Method for dust sampling (vacuum & wipes) at building with metals & pesticide/ herbicide impacts.
Community Environmental Health Resource Center	Lead Dust Background Materials	Undated	Instructions for sampling presented in checklist format; specific to dust related to lead paint.
US Department of Housing and Urban Development (US HUD)	Study of HUDs Risk Assessment Methodology in Three U.S. Communities	June 2006 (revised)	Evaluates sampling protocols and lead guidelines relative to their likelihood of predicting elevated blood lead.
US HUD	Guidelines for the Evaluation and Control of Lead-Based Paint Hazards in Housing	July 2012 (second edition)	Chapter 5 includes dust sampling methods; Appendix 13.1 contains detailed methods for wipe sampling (lead only).
US EPA Office of Pollution Prevention and Toxics	Residential Sampling for Lead: Protocols for Dust and Soil Sampling Final Report	March 1995	Dust sampling methods for wipes & vacuum, including details on laboratory preparation analysis (focus is on lead).
US EPA Office of Solid Waste and Emergency Response	Guidance for the Sampling and Analysis of Lead in Indoor Residential Dust for Use in the Integrated Exposure Uptake Biokinetic (IEUBK) Model	December 2008	Methods for collecting and analyzing residential dust lead data for use with the IEUBK model in Superfund risk assessments (lead only).
Verein Deutscher Ingenieure (VDI)	German Protocol VDI 4300, Part 8. Measurement of Indoor Air Pollution: Sampling of House Dust	June 2001 (withdrawn August 2012; revised December 2013 version [Part 11] only addresses airborne particulates)	Describes sampling conditions for house dust, including vacuum and wipe sampling for a variety of chemicals.

3.2.1 Wipe Sampling (Surface Area Measurement)

Wipe sampling is generally more cost-effective and simpler than vacuum sampling, and requires no special equipment. However, wipe samples can be obtained only from hard surfaces, and applying a consistent site sampling procedure among multiple individuals can be a challenge. Depending on the chemical being analyzed, researchers have used common cleaning wipes readily available at many stores; however, it is essential that the wipes used do not have elevated concentrations of the chemicals under consideration and that they do not contain substances that may interfere with the analysis.

A wipe sampling method was developed by the US Department of Housing and Urban Development (US HUD, 2012) for determination of lead. According to this method, wipes have to be of a single ply, cannot easily tear, must have low background levels of the target analytes, be digestible in a laboratory with a consistent recovery percentage and remain moist during sample collection. Other requirements may exist depending on the nature of the analyte and laboratory analysis to be performed.

Sampling is typically accomplished by wiping a known area in an "S" pattern using as much surface area of the wipe as possible while applying even pressure, then folding the wipe in half (sampled side in) and repeating the "S" motion perpendicular to the first pass (US HUD, 2012). Minor variations to the wiping procedure exist among protocols and are generally presented as a guideline to achieve consistent sampling throughout a site. Powder-free nitrile gloves are required to prevent contamination of the sampling area, and samples should be immediately stored in laboratory-approved containers.

3.2.2 Vacuum Sampling (Bulk Dust Measurement)

While generally more difficult and costly than wipe sampling, vacuum sampling can be applied to soft surfaces, such as upholstery, carpeting, rugs or drapes. Additionally, larger sample areas can be investigated as compared with those sampled with wipes, dust can be screened by particle size and, if the weight of dust collected is known, the chemical concentration in dust can be obtained in addition to the surface loading. The results of vacuum sampling are affected by several variables, including the type of carpet, age, condition, location in the residence and cleaning frequency (Lioy *et al.*, 2002).

Specialized units exist specifically for indoor dust sampling (e.g., CS3 Inc.'s HVS3 [High Volume Small Surface Samplers]), but some models of standard household vacuum cleaners can be adapted to obtain floor dust samples. Vacuum sampling requires an air pump apparatus with a removable filter, capable of maintaining a relatively constant flow rate. Sampling is performed by vacuuming in one direction across a known area at a constant rate. Vacuuming is repeated perpendicular to the previous pass and again in the opposite direction to the first pass. Filters, sampling canisters and/or bags are then removed from the unit and stored in Ziploc bags or containers provided by the analytical laboratory.

Vacuum sampling can be time consuming, as equipment should be regularly calibrated and cleaned (which may involve disassembling part of the vacuum) and the filter replaced after each sample. Overloading the vacuum can also be an issue, particularly for filter-based systems, as this can result in preferential sampling of smaller, less dense particles (US EPA, 1995) or may require multiple filters for a single sample area. Occasionally, there may be insufficient sample mass to meet analytical requirements, and sample collection from a larger area may be required. The actual area sampled should be calculated and provided in the report, in addition to the sample mass obtained. Where multiple samples are obtained in buildings, information related to each sampling area should be provided separately and shown on a figure.

Use of household vacuum cleaners can pose some problems, since each type has different design and particle collection/retention characteristics. Although the determination of particle size fraction distribution may be unreliable, household vacuums can provide information on the presence or absence of elements as well as the multi-element signature of the dust (Rasmussen *et al.*, 2001; Lioy *et al.*, 2002). Particle size fractionated mass distributions can be determined by use of a well-characterized standard vacuum (Lioy *et al.*, 2002). Several studies have relied on use of household vacuum cleaners with consistent results (Rasmussen *et al.*, 2001; Fan *et al.*, 2010).

At the current time, it would be inadvisable to collect household vacuum bags without knowledge of the areas sampled by the homeowner. Ideally, information should be gathered regarding the characteristics and activities in the home, such as home age, cleaning patterns and construction activities (Lioy *et al.*, 2002).

The specific type of sampling system used may depend to some extent on the desired particle size range, as filter-based systems may be more effective at sampling ultrafine particles, whereas systems with a sampling canister or bag can collect larger particles more readily. The flow rate of the vacuum, the collection efficiency, the precision and the size or portability of the unit may also be considerations (US EPA, 2008b).

Vacuum sampling methods used in the CHDS (e.g., Rasmussen *et al.*, 2011) were based on the German vacuum protocol (VDI, 2001). McDonald *et al.* (2010) provided information on wipe sampling methods used in the CHDS for lead and cadmium. McDonald *et al.* (2011) provided information related to wipe sampling for metals other than lead.

3.2.3 Wipe Versus Vacuum Sampling

Vacuum and surface wipe sampling can both be useful for evaluation of chemicals in indoor dust. One of the main advantages of surface wipe sampling is that chemical concentrations in units of $\mu g/m^2$ are better understood with respect to exposure potential. However, vacuum dust sampling can provide bulk concentrations as $\mu g/g$, which may be compared with concentrations in soil at a contaminated site. Additionally, larger dust samples can be obtained through vacuum sampling, which can allow for more extensive and sophisticated analysis techniques to be employed on the sample (such as speciation and synchrotron studies) (Rasmussen *et al.*, 2008, 2011; MacLean *et al.*, 2011).

In a workshop held by Health Canada in 2009, participants from academia, government and consulting companies provided feedback on the advantages and disadvantages of the two methods. Table 2 below provides a summary of some of the possible advantages and disadvantages of these sampling methods.

Sampling Method Considerations	Vacuum Sampling (Bulk Dust Measurement)	Surface Wipe Sampling (Surface Area Measurement)
International acceptance	European regulatory agencies are using this approach in addition to the USA (HVS3).	USA regulatory agencies are using this approach for lead (US EPA, US HUD, World Trade Center Indoor Air Task Force Working Group [WTCWG]).
Application for multiple chemicals	Published information on bulk dust concentrations is useful for multi-element studies (Rasmussen <i>et al.</i> , 2001).	Extensive wipe sampling results for lead are published in the literature; but not for other chemicals.
Cost and level of difficulty	More costly and time consuming to perform.	Easy and inexpensive to perform.
Particle size considerations	Able to include particle size considerations (Lioy <i>et al.</i> , 2002).	Typically cannot include particle size analysis.
Relationship to outdoor concentrations	Able to perform indoor-outdoor comparisons with soil, assuming that same particle size fractions selected.	Cannot perform indoor-outdoor comparisons because of difficulties collecting a surface wipe sample for soil or isolating specific particle size fractions.
Standardization of sampling protocol	Possible to apply standard sampling protocols consistently because minimal sample requirements can typically be met. Adds credibility and comparability to method.	Sometimes sampling protocol applied in a non-standard way as a result of lack of sample. (Expanding the area is recommended when a minimal sample is available.)
Spatial variability and sample size	Collection of composite bulk dust sample can avoid significant room-to-room variability, and since dust from multiple rooms can be combined, collection of an adequate sample size with reasonable detection limits is possible. Sampling can be conducted for individual rooms as well when information on spatial variability is needed.	Room-to-room variability can skew wipe sample results. However, information about individual rooms may be advantageous for examining spatial variability within residence.
Influence of housekeeping practices	Less likely to be influenced by recent housekeeping practices and thus may be more likely to provide information that would assist in source identification.	More likely to be influenced by recent housekeeping and may be more difficult to assist in source identification.

Table 2: Possible Advantages and Disadvantages of Vacuum Versus Surface Wipe Sampling Methods*

*Information based on opinions expressed at October 20th, 2009, workshop hosted by Health Canada's Contaminated Sites Division

3.2.4 Recommendations for Dust Sampling

As this document does not provide a recommended protocol for dust sampling, protocols developed by other agencies can be consulted when developing a site-specific protocol. Locations of samples should be identified prior to sampling and should be clearly stated in the HHRA report. The decision to sample high-use areas of the building or to conduct composite sampling should be identified in advance, and the supporting rationale and uncertainty discussion should be presented in the report.

Before conducting dust sampling, the project requirements and scope need to be evaluated, as well as the relative amount of hard versus soft surfaces present in the building(s) of interest, as these may affect the selection of sampling methods. In general, the following recommendations are made:

• Depending on the sample size requirements, it may be prudent to request that the building not be cleaned for one week before sampling to ensure that a sufficient sample size can be obtained.

- Vacuum sampling should be used for soft surfaces, using appropriate equipment.
- Either wipe sampling or vacuum sampling can be used for hard surfaces; if the concentration by weight is desired then vacuum sampling may be more appropriate.
- All sampling should be conducted in accordance with a comprehensive protocol and quality control/quality assurance program.
- Detection limits (and DSC) for each chemical should be identified in advance of sampling.
- A sampling plan should be submitted for review by the custodian prior to sampling.
- Custodian should confirm whether approvals are required in advance of sampling (e.g., if residences are sampled).

3.3 Laboratory Analysis

Specifications on laboratory analysis are not provided in this document. It is recommended that the analytical method be identified in advance of collection, as this may influence the amount of sample required. If one of the goals of the assessment is to identify how much contaminated soil is tracked in, then it may be prudent to identify whether the laboratory methods used would enable comparison of the concentrations in soil and indoor dust.

Various laboratory methodologies for the chemical analysis of vacuum dust are provided in published papers (e.g., Adgate *et al.*, 1995; Bai *et al.*, 2003; Fan *et al.*, 2010; Freeman *et al.*, 1997; Lewis *et al.*, 1999; Rasmussen *et al.*, 2008, 2011, 2013; Reynolds *et al.*, 1997; Rudel *et al.*, 2001), and methodologies for the analysis of surface wipes are also published (Adgate *et al.*, 1995; Bai *et al.*, 2003; Dufresne *et al.* 2011; Lioy *et al.*, 2000; McDonald *et al.*, 2010, 2011; Reynolds *et al.*, 1997).

The laboratory performing the chemical analyses should be certified by an accredited laboratory organization, such as the Canadian Association for Laboratory Accreditation or a similar organization like the Programme d'accréditation des laboratoires d'analyse of Quebec. If standardized methods have not been developed, then use of a research laboratory may be warranted, but in that case the analytical protocol and quality assurance procedures should be thoroughly documented and justified.

3.4 Quality Assurance and Quality Control

Quality control (QC) samples should be collected and analyzed to assess precision and/or bias of the sampling and analysis process. The collection and analysis of appropriate QC samples, as part of a quality assurance (QA) program, can help ensure that the quality of the data collected is known and that it meets a project's data quality objectives (DQOs). To establish the DQOs, the degree of certainty that is required should first be determined. QA consists of those activities that ensure that a defined standard of data quality with a stated level of confidence is met.

This document will not provide specific guidance on QA/QC requirements, and the information listed in this section is not a comprehensive guide. The report should identify the QA/QC procedures used as well as interpretation of the results (e.g., of duplicate or blank samples) and the associated uncertainty. A project's DQOs should be defined at the outset of the project to establish acceptable levels of data precision, bias, representativeness, completeness, comparability and detection limits. QA procedures, including the collection of field QC samples and their required frequencies, should be established in order to monitor whether the DQOs are being met. QC results should be reviewed and interpreted on an ongoing basis and the QA procedures modified as necessary. At project completion, an evaluation of the project data quality should be presented in a report.

The number of samples required for assessing loading or concentration in a location is dependent on the total area being evaluated and is often large for indoor settled dust analysis. Several duplicate samples are often required, as variability among samples can be high (US HUD, 2006) and neither the wipe nor vacuum samples collected on filters can be sub-sampled for different analyses (bulk vacuum samples can be sub-sampled). Other techniques, such as composite sampling, may or may not be available, depending on the chemical in question and the method of analysis. Additionally, many laboratory procedures for dust analysis require multiple additional blank and fortified samples in order to verify appropriate recovery levels from the sample matrix (US EPA, 1995).

The surface sampled may also affect the number of samples required. It has been shown that areas such as window sills can have higher variability than floors (US HUD, 2006), and a single samples may therefore not be representative of exposure. Compared with sampling of other media, such as soil or water, indoor settled dust sampling requires more consideration of the types of surface to be sampled in order to obtain meaningful results. Detailed documentation of the sample location as well as the surface type and use should be provided in the report.

Field quality control samples indicate the precision (random variation) and bias (systematic error) associated with field sampling. The types of sample that may be collected and analyzed to quantify the data precision or bias include, but are not limited to, the following:

- Field replicates—split of the same sample that measures sampling precision;
- Blank samples—indicate whether samples have been contaminated during the sampling, shipping or analysis stages. Typical blank samples include trip, field and equipment blanks. These may be wipes or vacuum cartridges that are not used for sampling but otherwise subjected to the same handling as the samples of interest.

Samples should be labelled with sample number and location for each sampling period. Samples prepared in the field should be stored at an appropriate temperature, as determined by the testing laboratory, and submitted to the testing laboratory within acceptable holding times (defined for each chemical with the testing laboratory during the scoping process). The wearing of gloves, proper decontamination and other QA/QC procedures should be respected and documented.

Laboratory QA/QC samples indicate the precision (random variation) and bias (systematic error) associated with laboratory analysis. The types of samples that may be analyzed to quantify the data precision and bias are as follows:

- Laboratory duplicates, which are splits of the same sample, taken in the laboratory and analyzed using the same procedures. The samples are used to ensure that method performance is within accepted limits;
- Fortified samples, which are samples to which a known quantity of a chemical has been added. This can include blanks or other samples being analyzed. The sample is run on the analytical instrument to assess instrument bias and to determine whether the sample matrix has an influence on the quality of the result; and,
- Reference standards (also known as "standard" or "certified" reference materials), which are samples prepared by a laboratory or an outside body and which contain specified concentrations of chemicals with a specified margin of error. The reference standards are used to calibrate analytical instruments.

4.0 INCLUSION OF COPCS IN INDOOR DUST IN AN HHRA

Inclusion of COPC in indoor dust as an exposure medium in an HHRA can be achieved by using a method consistent with that outlined in HC's DQRA guidance (HC, 2010a). The exposure assessment, which estimates the extent to which individuals are exposed to COPCs in dust, involves calculation of exposure to COPCs in the dust through both ingestion and dermal contact. The following sections provide some considerations relevant to indoor dust media at contaminated sites; they expand on the discussion of HHRA related to indoor dust found in the DQRA and preliminary quantitative risk assessment (PQRA) guidance documents (HC, 2012a; 2010a).

4.1 Selecting COPCs in Dust

Typically, at a contaminated site, concentrations of chemicals in soils are screened against guidelines or criteria developed for specific land uses by the CCME or provincial jurisdictions. Screening chemicals in dust to identify which substances are COPCs is more complicated than screening of chemicals in soil, as there are no environmental guidelines for the protection of human health for settled dust. As discussed above, COPCs from contaminated soils may impact indoor dust, and therefore the COPCs in soils would be used to target the chemicals at a contaminated site that may impact indoor dust and therefore need to be considered. This guidance has been prepared for application at historically contaminated sites and does not cover sites with industrial impacts on outdoor air and indoor dust. This guidance cannot provide prescriptive advice on when dust should be considered as an exposure medium, versus assessment of COPCs in soil without separate analysis of dust. This decision is the responsibility of the risk assessor, who would provide the rationale on a site-specific basis. Inclusion of dust as an exposure medium depends on the complexity of the risk assessment and would be considered in more complex assessments when human direct exposure to soil/dust is a critical pathway and the ingestion of dust and soil could be considered separately.

Further screening of soil COPCs is recommended whereby measured dust concentrations are compared with:

- 1. Human health-based dust screening concentrations (DSC_{HH}); and
- 2. Baseline dust concentrations (if available) that are typical of concentrations found in Canadian homes that are not impacted by contaminated sites.

This two-step approach is presented to meet the variable needs of contaminated site management. Sometimes the project requires concentrations based on human health effects to determine whether a dust concentration exceeds a level of potential human health significance. Alternatively, the project may require that dust concentrations be screened against available baseline concentrations to establish what is typical or atypical for a site. Project needs may dictate that both types of information are required for screening and interpretation. The report should describe the method used to screen COPCs, with references as appropriate.

 DSC_{HH} can be developed on a site-specific basis prior to collection of samples and may be used for screening purposes provided that the values are interpreted appropriately by a trained risk assessment practitioner. The trained risk assessor should be aware of the assumptions and limitations of the DSC_{HH} , which can be calculated using the approach outlined in this report. If a COPC that is present at elevated concentrations in soils at the site is measured in dust at concentrations consistent with normal baseline concentrations for indoor settled dust in Canada, then that chemical would generally not be considered a COPC in dust. It is important that the statistical interpretation of such data be presented in the report and justified. Some information regarding assessment of baseline dust concentrations is presented in this report, and other information may be obtained from the published literature.

The statistics to be used to describe people's exposure to COPCs in indoor dust will depend on the quantity, quality and variability of available sampling data. It is important to apply a conservative estimate of exposure, based on the results of indoor dust sampling. Any decision to include dust in an HHRA needs to consider the concentrations of contaminants in dust to which people will be exposed (e.g., people are generally exposed to dust in the middle of a room, rather than dust in an unused attic space). The data used to estimate the exposure concentrations should be adequately described, and figures of sampling locations in the buildings should be included. Because of the intrusive nature of indoor sampling, it is often the case that fewer samples are available for settled dust measurements than for investigations of outdoor media. Consequently, upper-bound estimates of concentrations are often used in risk assessments, so that risks are not underestimated. For instance, typically one would use either the maximum or the 95th percentile of the data distribution, rather than mean concentrations.

The two-step approach for screening chemicals in dust is adapted from the CCME (2006) protocol, with further revisions based on work conducted by the WTCWG (2003). Deviations from the CCME (2006) approach include not assessing the background estimated daily intake to which a person would be exposed. Additionally, the concept of "geological background" dust concentrations was considered inapplicable and was replaced with a discussion of baseline dust concentrations.

The method for the calculation of the DSC_{HH} is provided in Appendix I.

4.1.1 Screening Against DSC_{ин}—Surface Area and Bulk Dust

This section presents human health effects-based dust concentrations as the **first step** for screening. This step should be considered if the project requires a determination of whether a dust concentration exceeds a level at which there is a potential for adverse human health effects.

Human health effects-based dust concentrations and baseline dust concentrations may be presented as both surface area (units of $\mu g/m^2$) and bulk (units of $\mu g/g$) measurements, because both are considered of relevance to human health.

Dust screening concentrations expressed in units of surface area will ensure that indoor environments that are particularly dusty will be given higher priority than environments that are not dusty. For example, two commercial buildings could have concentrations of a chemical in dust of 150 μ g/g and be treated similarly if only dust screening concentrations expressed as bulk dust concentrations are compared; however, if one of the buildings has a dust level of 1000 μ g/m² and the second has a dust level of 100 μ g/m², then the first building may be of higher priority from a health risk assessment perspective. Providing dust screening concentrations on a surface area basis therefore allows for the consideration of possibly greater exposures from dustier environments.

On the other hand, expression of dust screening concentrations as bulk dust concentrations can ensure that potential indoor sources are adequately considered without the confounding effects of recent cleaning. For example, depending upon cleaning habits, a residence with lead-based paint on window sills could have a surface area concentration that is similar to a residence without lead-based paint. However, if lead-based paint is flaking, measurement of dust as bulk dust concentrations may allow for identification of potential sources of exposure (i.e., high lead concentrations) even if the building is not particularly dusty. Thus, providing dust screening concentrations on a bulk dust concentration basis may allow for proper identification of potential sources regardless of cleaning habits.

One example of how both measurements may be important in different circumstances is a study conducted by Yiin *et al.* (2000), examining residential dust lead concentrations and their relationship to blood lead in preschool children. The findings suggest that dust measurements would be more relevant for health risk assessments when presented as dust concentrations in $\mu g/g$ for carpets but also as lead loadings in $\mu g/m^2$ on hard surfaces. In this study, blood lead concentrations correlated the highest with each of these measurements for the respective surface type.

For these reasons, dust screening concentrations may be developed as both surface area concentrations (i.e., $\mu g/m^2$) and bulk dust concentrations (i.e., $\mu g/g$).

Exposure calculations that may be used to derive DSC_{HH} are provided in Appendix I.

4.1.2 Screening Against Baseline Dust Concentrations

This section describes the use of available baseline dust concentrations as the **second step** for screening dust concentrations. This step should be considered if the project requires that dust concentrations be screened against available baseline concentrations to interpret what is typical or atypical for a region.

The concept of background, or baseline, concentrations of elements in soil versus dust differs significantly between the two media. This becomes apparent when comparing the background soil terms used in soil quality guidelines versus any measure of background dust for screening purposes.

Unlike soil, the concept of "natural" or "geological" background may not be applicable to indoor dust, since elements in dust stem from a variety of anthropogenic, natural, indoor and outdoor sources. Dust in the indoor environment is a complex matrix stemming from a variety of sources, to which inhabitants are exposed as a whole. Therefore, to isolate the portion of background dust that is considered to be natural/geological may not be particularly meaningful for dust exposure. Estimates of concentrations of chemicals in typical, urban background or "baseline" dust provide a more meaningful point of comparison for the interpretation of indoor dust samples at contaminated sites. The use of an arithmetic mean baseline dust concentration is recommended for comparison when sufficient data are available, and when the data are normally distributed, since it is generally a conservative estimate of central tendency. Valid arguments may exist for selection of a different baseline dust concentration statistic, such as the median, 95% upper confidence limit of the mean, 95th percentile or maximum, and the report should provide a clear rationale for the statistic selected.

High-quality local or regional baseline dust data, where available, may be used for the interpretation of dust concentrations. Specific guidance is not provided for a dust sampling plan to collect regional data, as a statistical analysis would be required on a site-specific basis, as well as identification of areas not impacted by the site. In the absence of high-quality local or regional dust data, national baseline dust data published in the CHDS (HC, 2012b; see Table 3, below, for examples) may be useful for the interpretation of dust concentrations. Other sources can be considered when preferred sources are not available. Data quality can be evaluated by considering such factors as the chemicals tested, dust measurement made, particle size assessed, study design, sampling design, dataset quality, and scale and representativeness of the data.

Canadian studies on baseline concentrations of chemicals in dust exist and could be useful for interpreting dust sampling results from contaminated sites. The CHDS (HC, 2012b) measured baseline (typical urban background) dust concentrations from a combination of indoor, outdoor, anthropogenic and natural sources. The results of this urban, cross-country study are statistically significant and nationally representative.

Table 3 provides examples of some published and peer-reviewed sources of Canadian baseline dust data and summarizes key study considerations for use on a comparative basis. Most of the available data are from the CHDS. Additional local and regional Canadian baseline dust data may be available for comparison and should be considered on a project-specific basis.

	Rasmussen <i>et al.</i> (2001)	Fan e <i>t al.</i> (2010)	Rasmussen <i>et al.</i> (2011)	McDonald e <i>t al.</i> (2011)	Rasmussen <i>et al.</i> (2013)
Chemicals tested	Ag, As, Be, Cd, Cr, Co, Cu, Hg, Mn, Mo, Ni, Pb, Sb, Se, Sr, Tl, U, Zn (and others).	Parabens, triclosan, methyl triclosan	Pb	As, Cd, Cr, Cu, Ni, Pb, Sb	As, Cd, Cr, Cu, Ni, Pb, Zn
Study	Ottawa pilot study	Data subset from Canadian House Dust Study (CHDS)	Dataset from entire CHDS	Data subset from CHDS	Dataset from entire CHDS
Dust measurement	Bulk dust (µg/g)	Bulk dust (µg/g)	Bulk dust (µg/g)	Surface area dust (µg/m²)	Bulk dust (µg/g), element loading (surface area dust; µg/m ²) and element daily loading rate (µg/m ² -d)
Particle size assessed	100–250 μm	<80 µm	<80 µm	Not applicable.	<80 µm (bulk dust); <300 µm (surface area dust & element daily loading rate)

Table 3: Examples of Published Canadian Baseline Dust Data

	Rasmussen <i>et al.</i> (2001)	Fan <i>et al.</i> (2010)	Rasmussen <i>et al.</i> (2011)	McDonald <i>et al.</i> (2011)	Rasmussen <i>et al.</i> (2013)
Study design considerations	Urban. 50 private dwellings from Ottawa built between 1893 and 1987. Random sampling design.	Urban. 63 single-family dwellings from a Canadian city with population >100 000. Random sampling design.	Urban. 1025 single-family dwellings from 13 Canadian cities with population >100 000. Three-stage stratified random sampling design.	Urban. 222 single-family dwellings from 3 Canadian cities with population >100 000. Random sampling design.	Urban. 1025 single-family dwellings from 13 Canadian cities with population >100,000. Three-stage stratified random sampling design.
Sampling design	Composite fresh dust samples collected by participants from bare floors and carpets following a protocol designed to capture recent surface dusts and avoid longer- term sinks, and using participant vacuums.	Composite fresh dust samples collected from bare floors and carpets following the German vacuum protocol, Verein Deutscher Ingenieure (VDI), using Pullman-Holt High-Efficiency Particulate Air (HEPA) vacuum sampler. Household vacuum dust also obtained from vacuum systems used by participants.	Composite fresh dust samples collected from bare floors and carpets following the German VDI vacuum protocol, using Pullman-Holt HEPA vacuum sampler.	Wipe samples collected from non- carpeted floors only in various rooms (main entry area, kitchen, living and family rooms, adult and child bedrooms and children's primary play area) using Ghost Wipes following the American Society for Testing and Materials (ASTM) method E-1792.	Composite fresh dust samples collected from bare floors and carpets following the German VDI vacuum protocol, using Pullman-Holt HEPA vacuum sampler. Element loadings and loading rates were calculated from questionnaire data.
Dataset quality	48 samples. Not statistically significant.	63 samples. Not statistically significant.	1,025 samples. Statistically significant.	932 samples. Not statistically significant.	1025 samples. Statistically significant.
Scale of data	Local Ottawa, Ontario, study.	National but not nationally representative.	Nationally- representative.	National but not nationally representative.	Nationally representative.

	Rasmussen <i>et al.</i>	Fan <i>et al.</i>	Rasmussen <i>et al.</i>	McDonald <i>et al.</i>	Rasmussen <i>et al.</i>
	(2001)	(2010)	(2011)	(2011)	(2013)
Analytical methods used	Acid digestion and total element concentrations determined by cold- vapour AAS (Hg) and ICP-MS (all other elements).	Sonication extraction of samples and total chemical concentrations determined using GC-MS.	Bioaccessible Pb measured through dilute HCI extraction and modified European Toy Safety Protocol. Pb _T determined by 4-acid digestion and ICP-OES and ICP-MS.	Samples digested according to modified ASTM method E 1644, and concentrations determined by ICP-MS.	Total concentrations of Pb, Zn, Cu, Cd, Ni, As (62 samples) & Cr (some samples) determined by 4-acid digestion and ICP-OES or ICP-MS. For other samples, total concentrations of As & Cr determined by INAA.

Acronyms:

AAS: atomic absorption spectrometry

GC-MS: gas chromatography-mass spectrometry

HCI: hydrochloric acid

ICP-MS: inductively coupled plasma-mass spectrometry

ICP-OES: inductively coupled plasma-optical emission spectrometry

INAA: instrumental neutron activation analysis

 Pb_{T} : total lead concentration

4.1.3 Selection of COPCs

COPCs in indoor dust should be determined by screening against DSC_{HH} and baseline concentrations. If a chemical at a contaminated site is considered a COPC in soil (by comparison against Canadian soil quality guidelines for the protection of human health (SQGs)), it should automatically be considered a COPC in dust, if dust is determined to be a media of interest. If no baseline concentrations are available, chemicals may be screened only against DSC_{HH} .

If the chemical concentrations are below baseline, the chemical would not be brought forward into the risk assessment for dust exposure alone. However, the data may be used as part of a multi-media exposure assessment. Dust would not be required to be remediated below baseline concentrations, because these are representative of the general public's exposure unaffected by contamination.

4.2 Receptor Characterization

Receptor characteristics are provided in the Health Canada PQRA Part 1 document (HC, 2012a) for risk assessments of contaminated sites on federal lands. The input parameters for receptor characteristics used in the assessment should be documented in a table in the report to ensure that the assessment is transparent and the calculations can be confirmed by a technical peer reviewer. For site-specific HHRAs, any site-specific behaviour should be addressed in the HHRA report, with justification provided.

Receptor characteristics provided as an example for the purpose of derivation of the DSCs are summarized in Table 4; however other values may be used based on site-specific exposure factors. Rationale for all receptor characteristics should be documented in the risk assessment. Upper-bound estimates of dust ingestion may be used in risk assessments, so that risks are not underestimated, which may include use of the 95th percentile of the data distribution, rather than mean concentrations.

The PQRA Part 1 document (HC, 2012a) does not provide a value for ingestion of indoor dust, and this parameter has not been extensively quantified in the published literature. For the purpose of derivation of DSCs, this parameter is required. Exposure to dust via object-to-mouth behaviours cannot be quantified by mass-balance soil ingestion studies since its relative contribution to excreted trace elements cannot be distinguished from that of soil (Calabrese and Stanek, 1992a, 1992b; Stanek and Calabrese, 1992; Stanek *et al.*, 2012). In addition, mass-balance studies of soil ingestion have been shown to have highly variable results and are often fraught with uncertainties (Wilson *et al.*, 2013; Stanek *et al.*, 2012). Two potential sources of bias in mass-balance analysis are the lack of identification of non-soil sources that contribute to exposure (toothpaste, medication, food, paint, paper, crayons, etc.) as well as actual bioavailability and body retention of the elements used (Stanek *et al.*, 2012).

Ingestion of settled indoor dust has been quantified in few studies in the published literature. Ozkaynak *et al.* (2011) modelled soil and dust ingestion rates using a mechanistic method based on the US EPA's Stochastic Human Exposure and Dose Simulation model. For children aged 3 to 6 years, Ozkaynak *et al.* (2011) predicted mean dust ingestion rates of 20 mg/day via hand-to-mouth behaviour and 7 mg/day of dust via object to mouth, with a mean soil ingestion rate of 41 mg/day. They estimated mean total soil and dust ingestion rates for this age group as approximately 68 mg/day.

Using a different approach to estimate dust ingestion rates, Wilson *et al.* (2013) conducted a probabilistic assessment based on a literature review. The authors estimated mean indoor dust ingestion rates ranging from 2 mg/d for teenagers to 41 mg/d for toddlers. This compared with a range of mean soil ingestion rates from 1 mg/d for seniors to 20 and 23 mg/d for toddlers and children respectively in the same study. Combined soil and dust ingestion rates were reported to range from 4 mg/d for seniors to 61 mg/d for toddlers. These combined soil and dust ingestion rates are lower than soil ingestion rates adopted by most agencies. However, the combined soil and dust ingestion rate for toddlers was similar to that estimated by Ozkaynak *et al.* (2011), and the toddler and child soil ingestion rates are mechanistic, can be adjusted on a site-specific basis, can be modified into an hourly rate and are presented as a more realistic alternative to traditional mass-balance approaches. These values have been provided as an example (Table 4); however, other values cited from scientific literature may be used. Additional details, including values recommended for probabilistic assessment, are available in Wilson *et al.* (2013).

Receptor characteristics, such as body weight and exposed dermal surface area, which are typically used in risk assessment of contaminated sites, are provided as an example in Table 4 below. Other parameters were identified from the published literature or calculated in the equations presented in Appendix I. Alternative values may be used in an HHRA with supporting rationale, ensuring that the risk assessment provides a conservative estimate of risk. Receptor characteristics identified in the risk assessment should be summarized, with a rationale for the selection.

Generic Input Parameter	Value Selected	Reference
Body weight (BW)	Infant: 8.2 kg	Health Canada (2012a)
	Toddler: 16.5 kg	
	Child: 32.9 kg	
	Teen: 59.7 kg	
	Adult: 70.7 kg	
Dust ingestion rate (DIG)—residential	Infant: 0.038 g/day	Calculated from Eq. 7 (Appendix I) [†]
	Toddler: 0.041 g/day	
	Child: 0.031 g/day	
	Teen: 0.0022 g/day	
	Adult: 0.0025 g/day	

Table 4: Example Receptor Characteristic Parameters to Derive Dust Screening Concentrations*

Generic Input Parameter	Value Selected	Reference
Dust ingestion rate (DIG)—commercial	Infant: 0.028 g/day	Calculated from Eq. 7 (Appendix I) [†]
	Toddler: 0.027 g/day	
	Child: 0.021 g/day	
	Teen: 0.0014 g/day	
	Adult: 0.0014 g/day	
Surface area of fingers inserted into	Infant: 0.0013 m ²	Calculated from Eq. 2 (Appendix I)
mouth (SA _{fingers})	Toddler: 0.0015 m ²	
	Child: 0.0021 m ²	
	Teen: 0.0020 m ²	
	Adult: 0.0022 m ²	
Surface area of hands exposed to dust	Infant: 0.032 m ²	Health Canada (2012a)
(both hands; SA _{hands})	Toddler: 0.043 m ²	
	Child: 0.059 m ²	
	Teen: 0.080 m ²	
	Adult: 0.089 m ²	
Surface area of other body parts exposed	Infant: 0.146 m ²	Health Canada (2012a)
to dust (SA _{other})	Toddler: 0.258 m ²	
	Child: 0.455 m ²	
	Teen: 0.720 m ²	
	Adult: 0.822 m ²	
Skin adherence factor for dust on hands (SAF _{hands})	2 g/m²-day	US EPA (2008a)
Skin adherence factor for dust on other body parts (SAF _{other})	0.3 g/m²-day	US EPA (2008a)
Saliva extraction factor (SE)	0.5	WTCWG (2003) and Camann <i>et al.</i> (2000)
Sleep time (ST), residential only	Infant: 13 h/day	Health Canada (1995)
	Toddler: 10.5 h/day	
	Child: 9.9 h/day	
	Teen: 9.1 h/day	
	Adult: 8.4 h/day	

* Note that other values may be used with a detailed rationale (with citations)

[†] Rates are consistent with those from Wilson et al. (2013)

4.3 **Exposure Characterization Assumptions**

Most assumptions concerning exposure frequency and duration are based on best professional judgement, as there is a paucity of information in the published literature regarding typical exposure parameters. Where available, these values have been cited. For purposes of an HHRA for a federal contaminated site, the frequency and duration of site exposure should be based on the guidance presented in Table 5 unless, in the opinion of the risk assessor, alternative assumptions are more defensible. Justification for alternative assumptions should be provided and fully referenced. The reader is referred to the DQRA guidance (HC, 2010a) for a discussion of the implications of exposure amortization.

The recommended approach for estimation of exposure to chemicals in dust involves the estimation of exposure (in units of μ g/kg-d) from the oral and dermal routes. These exposures are then summed to provide an estimate of the total exposure that could arise from indoor dust.

Dust exposures can be estimated assuming equal exposure to hard and soft surfaces or can be modified for buildings in which there is expected to be more or less exposure to hard or soft surfaces. The fraction of indoor time spent in contact with hard versus soft surfaces can be estimated to be 50/50, notably on the basis of results from a method development and evaluation study conducted under the national CHDS (McDonald *et al.*, 2011), which showed that, on average, 56% of homes surveyed had carpeted floors and 44% were not carpeted. Additionally, it can be assumed that people contact settled dust on surfaces that are soft (e.g., couches, beds) and hard (e.g., tables) on what could be an equivalent basis. "Hard surfaces" are defined as those composed of non-porous materials, such as hardwood, linoleum, tile, laminate, etc., and primarily refer to floors but do not exclude other hard dust-bearing surfaces such as countertops, tables, hard toys, and window sills. "Soft surfaces" are defined primarily as floors covered with carpets and rugs but not excluding other soft dust-bearing surfaces such as sofas, pillows and beds.

Several simplifying assumptions are made to allow for estimates of exposure, including the assumption that dust concentrations remain constant and that dust is transferred to hands with every contact event.

These exposures relate solely to contact with settled dust; airborne dust is evaluated separately through inhalation exposure assessments.

Generic Input Parameter	Value Selected	Reference
Dust surface loading on horizontal hard surfaces (DSL _{hard})	0.052 mg/cm ²	Arithmetic mean calculated from Johnson <i>et al.</i> (2009)
Dust surface loading on horizontal soft surfaces (DSL _{soft})	0.139 mg/cm ²	Arithmetic mean calculated from Roberts <i>et al.</i> (1999) and Laxen <i>et al.</i> (1988) datasets
Exposure frequency (EF)	Residential: 1 (7 d/wk, 52 wk/yr)	Health Canada (2012a)
	Commercial: 0.71 (5 d/wk, 52 wk/yr)	
Exposure factor for transfer to body (EFB)	Residential: 1.8 hr/day (toddler)	Calculated from Eq. 4 (Appendix I)
	Commercial: 1.2 hr/day (toddler/adult)	
Exposure factor for transfer	Residential: 5.17 hr/day (toddler)	Calculated from Eq. 3 (Appendix I)
to fingertips (EFF)	Commercial: 3.36 hr/day (toddler)	
	1.9 hr/day (adult)	
Exposure time for contact with horizontal	Residential: 6.1 hr/day (toddler)	Calculated from Eq. 8 (Appendix I)
hard surfaces (ET _{hard})	Commercial: 4.0 hr/day (toddler/adult)	

Table 5: Example Exposure Parameters to Derive Dust Screening Concentrations*

Generic Input Parameter	Value Selected	Reference
Exposure time for contact with horizontal	Residential: 6.1 hr/day (toddler)	Calculated from Eq. 9 (Appendix I)
soft surfaces (ET _{soft})	Commercial: 4.0 hr/day (toddler/adult)	
Frequency of hand-to-mouth events (FQ)	Infant: 28 hr ⁻¹	Geometric mean calculated from Xue
	Toddler: 16 hr ⁻¹	et al. (2007)
	Child: 9.1 hr ⁻¹	
	Teen: 1.0 hr ⁻¹	
	Adult: 1.0 hr ⁻¹	
Fractional surface area of fingers	Infant: 0.08	Median values from AuYeung et al. (2008)
for mouthing activity (FSA _{fingers})	Toddler: 0.07	
	Child: 0.07	
	Teen: 0.05	
	Adult: 0.05	
Fraction of dust transferred from hard surfaces to body (FTSS _{HSB})	0.25	WTCWG (2003)
Fraction of dust transferred from hard	Infant: N/A	Estimated from Rodes et al. (2001) based
surfaces to hands (FTSS _{HSH})	Toddler: 0.7	on dermal transfer factors
	Child: 0.7	
	Teen: 0.4	
	Adult: 0.4	
Fraction of dust transferred from soft surfaces to body (FTSS _{SSR})	0.05	WTCWG (2003)
Fraction of dust transferred from soft	Infant: 0.14	Estimated from Rodes et al. (2001) based
surfaces to hands (FTSS _{SSH})	Toddler: 0.14	on dermal transfer factors
	Child: 0.14	
	Teen: 0.08	
	Adult: 0.08	
Hours spent in environment	Residential: 24 hr/day	Health Canada (2012a)
each day (HPD) [†]	Commercial: 8 hr/day	
Transfer coefficient (TC) [‡]	0.12 m²/hr	WTCWG (2003)
Time spent outdoors (TO),	Infant: N/A	Toddler and child based on US EPA
assumed for residential only	Toddler: 1.2 h/day	(2008a) and supported by Leech
	Child: 2.2 h/day	from Richardson (1997)
	Teen: 1.4 h/day	
	Adult: 1.4 h/day	

* Use of these values is outlined in Appendix I.

[†] Note that some values may be altered on a site-specific basis, with rationale.

 ‡ The transfer coefficient represents the rate of skin contact with the surface.

4.4 Concentrations of COPCs in Indoor Dust—Use of Appropriate Statistics

4.4.1 Use of Maximum Concentration Versus Mean Concentration

Different statistics of the collected dust sampling data can be used to represent the concentration of a chemical in dust from which exposure is estimated. If limited data are available, the maximum measured concentration is recommended for use in exposure calculations. If the sample size is deemed sufficient and the collected samples are considered representative of the dust concentrations affected by the site, it may be possible to use a statistic representing the central tendency of the data (e.g., mean or median) for each building to provide a more realistic estimate of the typical or average concentration for exposure and risk calculation. Conservative estimates of the average exposure (such as the 95% upper confidence limit of the mean or median + 2*MAD [median absolute deviation]) are commonly used to maintain conservatism in the risk assessment where data are limited. The specific approach used will depend on the quantity and quality of the data and the purpose of the risk assessment, as well as considerations such as exposure duration, and should be determined on a case-by-case basis with appropriate justification. Often a tiered approach is applied in which maximum concentrations are used for the initial screening, and more realistic exposure estimates used if initial estimates exceed the DSC_{HH} or baseline.

Use of the geometric mean is not recommended. The geometric mean is a statistical representation of the central tendency of data that are log-normally distributed. However, this statistic is not necessarily representative of the average concentrations to which people will be exposed. The arithmetic mean (or upper confidence limit of the mean) is generally a more appropriate representation of the average or typical concentration to which a person will be exposed for normally distributed concentration data and a conservative estimate for other distributions (US EPA, 2002). The median may be a suitable measure of central value for other distributions (Reimann and Filzmoser, 2000; Reimann *et al.*, 2008).

4.4.2 Incorporation of Data with Non-Detected Concentrations

In some cases, measured dust concentrations may be reported as less than detection limit. Analytical data sets frequently include both reported concentrations (detects) and reported inability to detect the chemical (non-detects). Consequently, the low end of the distribution of concentrations is (left) censored. Non-detectable concentrations do not necessarily indicate that a chemical is absent, but rather, they indicate that the chemical may be absent or it may merely be present at concentrations below the method detection limit or quantification limit. If a chemical is detected in some samples from a site, it is possible that it is also present at low concentrations in samples reported as non-detects.

When a dataset includes non-detect values, methods to calculate exposure concentration point estimates for reported non-detect concentrations can be used, such as those described in the 'Soil Characterization Guidance' section of the *CCME (2016) Guidance Manual for Environmental Site Characterization in Support of Environmental and Human Health Risk Assessment* (Volume 1). Regardless of the approach used to address non-detect values, appropriate justification and full references should be provided.

4.4.3 Data Requirements

The data should be provided in the risk assessment report, preferably including tabulated data for individual samples, as well as tables with statistical summaries of the data (max, mean, min, number of samples). Table 6 identifies site-specific parameters that are required for input in the equations in Appendix I.

Table 6: Site-Specific Input Parameters for HHRAs

Generic Input Parameter	Value Selected
Bulk dust concentration on horizontal surfaces (BDC)	Measured, site-specific value (µg/g)
Dust concentration from hard surfaces on an area-weighted basis (DCA _{hard})	Measured, site-specific value (µg/m²)
Dust concentration from soft surfaces on an area-weighted basis (DCA _{soft})	Measured, site-specific value (µg/m²)

Note: HHRA may use either approach (bulk or area-weighted) or both approaches

4.5 Toxicity Assessment

The HHRA includes a toxicity assessment, as described in the PQRA and DQRA guidance documents (HC, 2012a; 2010a). Toxicological reference values (TRVs) may be obtained from HC's Part II: Health Canada's Toxicological Reference Values and Chemical-Specific Parameters (2010b). Alternately, if values for specific COPCs are not available in HC (2010b), values from other regulatory agencies may be used, as specified in the PQRA and DQRA guidance documents. For further information please refer to HC (2012a; 2010a,b).

4.6 Bioavailability of COPCs in Indoor Dust

The DQRA guidance (HC, 2010a) provides a section on considerations related to the bioavailability of COPCs in soil as compared with the bioavailability in the study used to derive the TRV. This comparison may be done on a site-specific basis, as bioavailability is strongly influenced by the matrix in which the chemical is found. On the basis of available data, it is expected that the oral bioavailability of chemicals from dust would be similar to or higher than the bioavailability from soil, particularly when the dust originates from contaminated soil (Freeman *et al.*, 1995; Oliver *et al.*, 1999; Yu *et al.*, 2006; Ruby *et al.*, 1996, 1999; Rasmussen, 2004; Rasmussen *et al.*, 2008). Therefore, it is recommended that dust exposure estimates assume that the bioavailability of the chemical in the exposure medium is similar to that in the study used to derive the TRV unless site specific data are available. Specifically, a relative oral bioavailability factor of 1 can be assumed for most chemicals (i.e., the bioavailability from dust is assumed to be the same as the bioavailability from the exposure medium in the critical toxicity study), unless chemical-specific information is available.

Dust exposure estimates could theoretically be adjusted on a chemical-specific basis, with sufficient supporting rationale. Use of *in vitro* methods demonstrating adequate validation with *in vivo* data from a variety of sources would be preferred, and use of additional toxicodynamic/toxicokinetic information could be considered for chemicals for which the TRV is provided as a biomarker limit or absorbed/internal dose.

Dermal bioavailability from soil can be assumed to be representative of dermal bioavailability from dust. Although dermal relative absorption factors are specific to soil, they can be adopted for dust since there is a lack of any medium-specific information and because dermal absorption of chemicals from the two media is not expected to differ markedly.

4.7 Risk Characterization

Risk characterization is the HHRA phase in which the exposure and toxicity data are compared, as discussed in the PQRA and DQRA guidance documents (HC, 2012a; 2010a). DSCs are established using a back-calculation of the equations used to generate risk estimates.

To calculate DSCs, it is recommended that the guidance for soil quality guideline development (CCME, 2006) be followed, in which effects-based dust concentrations are based on 20% of the TRV for non-carcinogens, which is reflected as a target hazard quotient (target HQ) of 0.2, or, for carcinogens, on a target incremental lifetime cancer risk of 1×10^{-5} (1 in 100,000).

The CCME (2006) protocol for derivation of soil quality guidelines does not identify indoor dust as a separate medium. It identifies soil, air, drinking water, food and consumer products as media to which people may be exposed, allocating an equal portion of the TRV to each medium (e.g., 20% to each medium). For soil and dust, the soil ingestion rate used in the CCME (2006) protocol does not differentiate between exposure to soil and exposure to dust. When soil is considered at a contaminated site, there is a possibility that dust may also be impacted, and thus should also be considered. If dust has not been collected and analyzed at the site, soil and dust can be treated as the same exposure medium (by assuming soil concentrations are a surrogate for dust concentrations) and either the CCME (2006) soil ingestion rate or a combination of soil+dust ingestion rates (e.g Wilson *et al.* 2013) can be used in the risk assessment. If dust has been collected and analyzed at the site, it can be treated as a separate medium, and the dust ingestion rates published by Wilson *et al.* (2013) may be used in addition to the soil ingestion rates that were calculated specific to the 'soil ingestion pathway only' by Wilson *et al.* (2013)—when the values in that publication are suitable for the land use. Although CCME 2006 soil ingestion rates do not differentiate between soil and dust, methodology such as the one used in Wilson provides the ability to separate soil and dust ingestion rates.

Exposures to soil and dust (or sediment) are not expected to occur concurrently, and thus separate allocations to these media are not proposed. Assigning 20% of the TRV to set screening concentrations with regard to potential human health significance of any particular exposure medium is consistent with HC screening-level guidance (e.g., HC, 2012a). Therefore, an allocation of 20% for dust is recommended.

In a DQRA in which dust is one of the media assessed and exposure from all media plus background is incorporated, a target HQ of 1 may be appropriate. For further information please refer to HC (2010a).

5.0 CONCLUSIONS

The information provided in this supplemental guidance document is recommended for use in conjunction with other HC guidance on assessment of risk from contaminated sites, specifically the DQRA guidance (HC 2010a). Inclusion of dust as a separate exposure medium is anticipated in more complex risk assessments at sites where direct exposure to soil and dust are important exposure pathways and when a decreased amount of uncertainty is required for the purpose of risk management or evaluation of mitigation measures. HC, Contaminated Sites Division, may be contacted with questions on the assessment of indoor dust in a contaminated sites risk assessment. The guidance provided in this document is not intended to be used for any purpose other than for risk assessment of federal contaminated sites where indoor dust may be a significant exposure pathway.

The HHRA report should provide a section on calculation of DSCs for screening COPCs as well as assessment of the COPCs carried forward into the risk assessment. Equations provided in Appendix I may be used to derive DSCs. Those provided in Appendix II may be useful in calculating exposures and risks in an HHRA. These exposures and risks can then be summed with those from other exposure pathways to COPCs at the site (e.g., from other contaminated media). The guidance provided in the DQRA document (HC, 2010a) is also recommended for use in HHRAs which incorporate the dust exposure pathway.

6.0 REFERENCES

Abt E, Suh HH, Catalano P, Koutrakis P. 2000. Relative contribution of outdoor and indoor particle sources to indoor concentrations. *Environmental Science & Technology* 34: 3579–3587.

Adgate JL, Weisel C, Wang Y, Rhoads GG, Lioy PJ. 1995. Lead in house dust: relationships between exposure metrics. *Environmental Research* 70: 134–147.

Afshari A, Matson U, Ekberg LE. 2005. Characterization of indoor sources of fine and ultrafine particles: a study conducted in a full-scale chamber. *Indoor Air* 15(2): 141–150.

Agency for Toxic Substances and Disease Registry. 2002. Final Technical Report of the Public Health Investigation to Assess Potential Exposures to Airborne and Settled Surface Dust in Residential Areas of Lower Manhattan. New York City Department of Health and Mental Hygiene, and Agency for Toxic Substances and Disease Registry, U.S. Department of Health and Human Services, as a part of the World Trade Center Environmental Assessment Working Group. September 2002. Accessed November 2, 2012, at: www.atsdr.cdc.gov/asbestos/asbestos/types_of_exposure/WTC_FullReport.html

American Society for Testing and Materials (ASTM). 2010. *Standard Practice for Collection of Settled Dust Samples Using Wipe Sampling Methods for Subsequent Lead Determination*. ASTM E1728–10. Accessed November 2, 2012, at: www.astm.org/Standards/E1728.htm

ASTM. 2011. Standard Practice for Collection of Floor Dust for Chemical Analysis. ASTM D5438–11. Accessed November 2, 2012, at: www.astm.org/Standards/D5438.htm

ASTM. 2011. Standard Specification for Wipe Sampling Materials for Lead in Surface Dust. ASTM E1792–03. Accessed November 5, 2012, at: www.astm.org/Standards/E1792.htm

AuYeung W, Canales RA, Leckie JO. 2008. The fraction of total hand surface area involved in young children's outdoor hand-to-object contacts. *Environmental Research* 108: 294–299.

Bai Z, Yiin L-M, Rich DQ, Adgate JL, Ashley PJ, Lioy PJ, Rhoads GG, Zhang J. 2003. Field evaluation and comparison of five methods of sampling lead dust on carpets. *American Industrial Hygiene Association Journal* 64: 528–532.

Calabrese EJ, Stanek ES. 1992a. Distinguishing outdoor soil ingestion from indoor dust ingestion in a soil pica child. *Regulatory Toxicology and Pharmacology* 15: 83–85.

Calabrese EJ, Stanek EJ. 1992b. What proportion of household dust is derived from outdoor soil? *Journal of Soil Contamination* 1(3): 253–263.

California Department of Health Services. 2006. *Exposure Investigation Protocol Dust Sampling in Building 240. Zeneca/ Campus Bay, Richmond, Contra Costa Country, California*. CAD009123456. Revised April 3. Accessed November 2, 2012, at: www.ehib.org/cma/projects/ZenecaEl.pdf

Camann DE, Majumdar TK, Geno PW. 2000. *Evaluation of Saliva and Artificial Salivary Fluids for Removal of Pesticide Residues from Human Skin*. Southwest Research Institute, San Antonio, TX, and ManTech Environmental Technology, Inc., Research Triangle Park, NC. For US Environmental Protection Agency, National Exposure Research Laboratory, Research Triangle Park, NC. EPA/600/R00/041.

Canadian Council of Ministers of the Environment. 2006. A Protocol for the Derivation of Environmental and Human Health Soil Quality Guidelines. PN 1332. Accessed November 5, 2012, at: www.ccme.ca/assets/pdf/sg_protocol_1332_e.pdf

CCME. 2016. Guidance Manual for Environmental Site Characterization in Support of Environmental and Human Health Risk Assessment, Volume 1–4, Canadian Council of Ministers of the Environment, Winnipeg.

Community Environmental Health Education Resource Centre. Undated. *Lead Dust Background Materials*. Accessed November 2, 20112, at: <u>www.afhh.org/hps/leaddocs/cehrc_lead_dust_materials.pdf</u>

Cornelis C, Swartjes FA. 2007. Ontwikkeling van een geharmoniseerde methodiek voor beoordeling van gezondheidsrisico's door bodemverontreiniging in de Kempenregio: VITO, Mol, Belgium and RIVM, Bilthoven, the Nederlands. Cited in Van Holderbeke *et al.*, 2008.

Davis JJ, Gulson BL. 2005. Ceiling (attic) dust: a "museum" of contamination and potential hazard. *Environmental Research* 99(2): 177–194.

Dixon SL, Gaitens JM, Jacobs DE, Strauss W, Nagaraja J, Pivetz T, Wilson JW, Ashley PJ. 2009. Exposure of U.S. children to residential dust lead, 1999–2004: II. The contribution of lead-contaminated dust to children's blood-lead levels. *Environmental Health Perspectives*117(3): 468–474.

Dufresne A, Mocanu T, Viau S, Perrault G, Dion C. 2011. Efficacy of surface sampling methods for different types of beryllium compounds. *Journal of Environmental Monitoring* 13: 74–83.

Fan CW, Zhang JJ. 2001. Characterization of emissions from portable household combustion devices: particle size distributions, emission rates and factors, and potential exposures. *Atmospheric Environment* 35: 1281–1290.

Fan X, Kubwabo C, Rasmussen P, Jones-Otazo H. 2010. Simultaneous quantitation of parabens, triclosan, and methyl triclosan in indoor house dust using solid phase extraction and gas chromatography-mass spectrometry. *Journal of Environmental Monitoring* 12(10): 1891–1897.

Freeman G, Schoof R, Ruby M, Davis A, Dill J, Liao S, Lapin C, Bergstrom P. 1995. Bioavailability of arsenic in soil and dust impacted by smelter activities following oral administration in cynomolgus monkeys. *Fundamental and Applied Toxicology* 28(2): 215–222.

Freeman NCG, Stern AH, Lioy PJ. 1997. Exposure to chromium dust from homes in a chromium surveillance project. *Archives of Environmental Health: An International Journal* 52(3): 213–219.

Health Canada (HC). 1995. Assessment of Mercury Exposure and Risks from Dental Amalgam. Prepared by GM Richardson, on behalf of the Bureau of Medical Devices, Health Protection Branch, Health Canada. August 18. Accessed November 6, 2012, at: <u>http://publications.gc.ca/collections/Collection/H46-1-36-1995E.pdf</u>

HC. 2010a. Federal Contaminated Site Risk Assessment in Canada, Part V: Guidance on Human Health Detailed Quantitative Risk Assessment for Chemicals (DQRA_{CHEM}). Contaminated Sites Division, Safe Environments Directorate, Health Canada, Ottawa.

HC, 2010b. Federal Contaminated Site Risk Assessment in Canada, Part II: Health Canada Toxicological Reference Values and Chemical-Specific Factors, Version 2.0. Contaminated Sites Division, Safe Environments Directorate, Health Canada, Ottawa.

HC. 2012a. Federal Contaminated Site Risk Assessment in Canada, Part I: Guidance on Human Health Preliminary Quantitative Risk Assessment (PQRA), Version 2.0. Contaminated Sites Division, Safe Environments Directorate, Health Canada, Ottawa.

HC. 2012b. *The Canadian House Dust Study.* Accessed November 5, 2012, at: <u>www.hc-sc.gc.ca/ewh-semt/contaminants/dust-poussiere-eng.php</u> (last modified July 26, 2011).

Johnson DL, Hunt A, Griffith DA, Hager JM, Brooks J, StellaLevinsohn H, Lanciki A, Lucci R, Prokhorova D, Blount SL. 2009. Geographic patterns of non-carpeted floor dust loading in Syracuse, New York (USA) homes. *Environmental Geochemistry and Health* 31: 353–363.

Jones-Otazo HA, Clarke JP, Diamond ML, Archbold JA, Ferguson G, Harner T, Richardson GM, Ryan JJ, Wilford B. 2005. Is house dust the missing exposure pathway for PBDEs? An analysis of the urban fate and human exposure to PBDEs. *Environmental Science & Technology* 39: 5121–5130.

Kissel JC, Richter KY, Fenske RA. 1996. Factors affecting soil adherence to skin in hand-press trials. *Bulletin of Environmental Contamination & Toxicology* 56: 722–728.

Klepeis NE, Apte MG, Gundel LA, Sextro RG, Nazaroff WW. 2003. Determining size specific emission factors for environmental tobacco smoke particles. *Aerosol Science and Technology* 37: 780–790.

Lanphear B, Matte T, Rogers J, Clickner R, Dietz B, Bornschein R, Succop P, Mahaffey K, Dixon S, Galke W, Rabinowitz M, Farfel M, Rohde C, Schwartz J, Ashley P, Jacobs D. 1998. The contribution of lead-contaminated house dust and residential soil to children's blood lead levels. A pooled analysis of 12 epidemiologic studies. *Environmental Research* 79(Section A): 51–68.

Laxen DPH, Lindsay F, Raab GM, Hunter R, Fell GS, Fulton M. 1988. The variability of lead in dusts within the homes of young children. *Environmental Geochemistry and Health* 10: 3–9.

Leech JA, Wilby K, McMullen E, Laporte K. 1997. The Canadian Human Activity Pattern Survey: report of methods and population surveyed. *Chronic Diseases in Canada*. 17(3): 118–123.

Leech JA, Nelson WC, Burnett RT, Aaron S, Raizenne ME. 2002. It's about time: a comparison of Canadian and American time-activity patterns. *Journal of Exposure Analysis and Environmental Epidemiology* 12: 427–432.

Lewis RG, Fortune CR, Willis RD, Camann DE, Antley JT. 1999. Distribution of pesticides and polycylic aromatic hydrocarbons in house dust as a function of particle size. *Environmental Health Perspectives* 107(9): 721–726.

Li CS, Lin WH, Jeng FT. 1993. Size distributions of submicrometer aerosols from cooking. Environment International 19: 147–154.

Lioy PJ, Edwards RD, Freeman N, Gurunathan S, Pellizzari E, Adgate JL, Quackenboss J, Sexton K. 2000. House dust levels of selected insecticides and a herbicide measured by the EL and LWW samplers and comparisons to hand rinses and urine metabolites. *Journal of Exposure Analysis and Environmental Epidemiology* 10: 327–340.

Lioy PJ, Freeman NCG, Millette JR. 2002. Dust: a metric for use in residential and building exposure assessment and source characterization. *Environmental Health Perspectives* 110: 969–983.

Long CM, Suh HH, Catalon PJ, Koutrakis P. 2001. Using time—and size-resolved particulate data to quantify indoor penetration and deposition behaviour. *Environmental Science & Technology* 35: 2089–2099.

MacLean LCW, Beauchemin S, Rasmussen PE. 2011. Lead speciation in house dust from Canadian urban homes using EXAFS, Micro-XRF, and Micro-XRD. *Environmental Science & Technology* 45: 5491–5497.

Matz CJ, Stieb DM, Davis K, Egyed M, Chou B, Brion O. 2014. Effects of Age, Season, Gender and Urban-Rural Status on Time-Activity: Canadian Human Activity Pattern Survey 2 (CHAPS 2). *International Journal of Environmental Research and Public Health*. Feb; 11(2): 2108–2124.

Matz CJ, Stieb DM, Brion O. 2015. Urban-rural differences in daily time-activity patterns, occupational activity and housing characteristics. *Environmental Health*. 14: 88.

McDonald LT, Rasmussen PE, Chénier M, Levesque C. 2010. Wipe sampling methodologies to assess exposure to lead and cadmium in Canadian homes. *Proceedings of the Annual International Conference on Soils, Sediments, Water and Energy.* Manuscript 1122. Part II: Heavy Metals. Pages 51–66.

McDonald LT, Rasmussen PE, Chénier M, Levesque C. 2011. Extending wipe sampling methodologies to elements other than lead. *Journal of Environmental Monitoring* 13(2): 377–383.

Mølhave L, Schneider T, Kjærgaard SK, Larsen L, Norn S, Jørgensen O. 2000. House dust in seven Danish offices. *Atmospheric Environment* 34: 4767–4779.

Morawska L, He C. 2004. Particle concentration levels and size distribution characteristics in residential and non-industrial workplace environments. In: Morawska L, Salthammer T (eds). *Indoor Environment: Airborne Particles and Settled Dust*. Weinheim, Germany: Wiley-VCH: 190–224.

Morawska L, Salthammer T. 2004. Fundamentals of indoor particles and settled dust. In: Morawska L, Salthammer T (eds). *Indoor Environment: Airborne Particles and Settled Dust*. Weinheim, Germany: Wiley-VCH: 3–46.

Morgan MK, Sheldon LS, Croghan CW, Jones PA, Robertson GL, Chuang JC, Wilson NK, Lyu CW. 2005. Exposures of preschool children to chlorpyrifos and its degradation product 3,5,6-trichloro-2-pyridinol in their everyday environments. *Journal of Exposure Analysis and Environmental Epidemiology* 15: 297–309.

Oliver D, McLauglin M, Naidu R, Smith L, Maynard E, Calder U. 1999. Measuring Pb bioavailability from household dusts using an *in vitro* model. *Environmental Science & Technology* 33(24): 4434–4439.

Oomen AG, Lijzen JPA. 2004. *Relevancy of Human Exposure Via House Dust to the Contaminants Lead and Asbestos*. RIVM report no. 711701037. Accessed November 2, 2012, at: <u>http://rivm.openrepository.com/rivm/bitstream/10029/9070/1/711701037.pdf</u>

Ozkaynak H, Xue J, Zartarian VG, Glen G, Smith L. 2011. Modeled estimates of soil and dust ingestion rates for children. *Risk Analysis* 31(4): 592–608.

Paustenbach DJ, Finley BL, Long TF. 1997. The critical role of house dust in understanding the hazards posed by contaminated soils. *International Journal of Toxicology* 16: 339–362.

Rasmussen P, Subramanian KS, Jessiman BJ. 2001. A multi-element profile of housedust in relation to exterior dust and soils in the city of Ottawa, Canada. *Science of the Total Environment* 267: 125–140.

Rasmussen PE. 2004. Can metal concentrations in indoor dust be predicted from soil geochemistry? Canadian Journal of Analytical Sciences & Spectroscopy 49(3): 166–174.

Rasmussen PE, Beauchemin S, Nugent M, Dugandzic R, Lanouette M, Chénier M. 2008. Influence of matrix composition on the bioaccessibility of copper, zinc and nickel in urban residential dust and soil. *Human and Ecological Risk Assessment* 14(2): 351–371.

Rasmussen PE, Beauchemin S, Chénier M, Levesque C, MacLean LCW, Marro L, Jones-Otazo H, Petrovic S, McDonald LT, Gardner HD. 2011. Canadian House Dust Study: lead bioaccessibility and speciation. *Environmental Science & Technology* 45: 4959–4965.

Rasmussen PE, Levesque C, Chénier M, Gardner HD, Jones-Otazo H, Petrovic S. 2013. Canadian House Dust Study: populationbased concentrations, loads and loading rates of arsenic, cadmium, chromium, copper, nickel, lead and zinc inside urban homes. *Science of the Total Environment* 443: 520–529. Reimann C, Filzmoser P. 2000. Normal and lognormal data distribution in geochemistry: death of a myth. Consequences for the statistical treatment of geochemical and environmental data. *Environmental Geology* 39(9): 1001–1014.

Reimann C, Filzmoser P, Garret RG, Dutter R. 2008. Statistical Data Analysis Explained. J. Wiley and Sons: 51–55.

Reynolds SJ, Etre L, Thorne PS, Whitten P, Selim M, Popendorf WJ. 1997. Laboratory comparison of vacuum, OSHA, and HUD sampling methods for lead in household dust. *American Industrial Hygiene Association Journal* 58: 439–446.

Richardson GM. 1997. Compendium of Canadian Human Exposure Factors for Risk Assessment. Ottawa, Ontario: O'Connor Associates Environmental Inc.

Riley WJ, McKone TM, Lai ACK, Nazaroff WW. 2002. Indoor particulate matter of outdoor origin: importance of size-dependent removal mechanisms. *Environmental Science & Technology* 36: 200–207.

Roberts JW, Clifford WS, Glass G, Hummer PG. 1999. Reducing Dust, Lead, Dust Mites, Bacteria, and Fungi in Carpets by Vacuuming. *Archives of Environmental Contamination and Toxicology* 36: 477–484.

Roberts JW, Wallace LA, Camann DE, Dickey P, Gilbert SG, Lewis RG, Takaro TK. 2009. Monitoring and reducing exposure of infants to pollutants in house dust. *Reviews of Environmental Contamination & Toxicology* 201: 1–39.

Rodes C, Smith T, Crouse R, Ramachandran G. 1990. Measurements of the size distribution of aerosols produced by ultrasonic humidification. *Aerosol Science and Technology* 13: 220–229.

Rodes CE, Newsome JR, Vanderpool RW, Antley JT, Lewis RG. 2001. Experimental methodologies and preliminary transfer factor data for estimation of dermal exposures to particles. *Journal of Exposure Analysis and Environmental Epidemiology* 11: 123–139.

Ruby MV, Davis A, Schoof R, Eberle S, Sellstone CM. 1996. Estimation of lead and arsenic bioavailability using a physiologically based extraction test. *Environmental Science & Technology* 30(2): 422–430.

Ruby MV, Schoof R, Brattin W, Goldade M, Post G, Harnois M, Mosby DE, Casteel SW, Berti W, Carpenter M, Edwards D, Cragin D, Chappell W. 1999. Advances in evaluating the oral bioavailability of inorganics in soil for use in human health risk assessment. *Environmental Science & Technology* 33(21): 3697–3705.

Rudel RA, Brody JG, Spengler JD, Vallarino J, Geno PW, Sun G, Yau A. 2001. Identification of selected hormonally active agents and animal mammary carcinogens in commercial and residential air and dust samples. *Journal of the Air & Waste Management Association* 51: 499–513.

Stanek EJ III, Calabrese EJ. 1992. Soil ingestion in children: outdoor soil or indoor dust? Journal of Soil Contamination 1(1): 1-28.

Stanek EJ III, Calabrese EJ, Xu B. 2012. Meta-analysis of mass-balance studies of soil ingestion in children. *Risk Analysis* 32(3): 433–447.

US Environmental Protection Agency (US EPA). 1995. *Residential Sampling for Lead: Protocols for Dust and Soil Sampling. Final Report.* Pollution Prevention and Toxics, EPA #747-R-95-001.

US EPA. 1997. Exposure Factors Handbook. National Center for Environmental Assessment, Washington, D.C

US EPA. 1998. *Short Sheet: IEUBK Model Mass Fraction of Soil in Indoor Dust (MSD) Variable*. OSWER #9285. 7–34. EPA #540-F-00-008. Accessed November 2, 2012, at: <u>http://epa.gov/superfund/lead/products/ssmsdcol.pdf</u>

US EPA. 2002. Calculating Upper Confidence Limits for Exposure Point Concentrations at Hazardous Waste Sites. Office of Emergency and Remedial Response, Washington, DC. OSWER 9285.6–10. Accessed November 14, 2012, at: www.epa.gov/oswer/riskassessment/pdf/ucl.pdf

US EPA. 2005. *World Trade Center Residential Dust Cleanup Program. Final Report.* US EPA Region 2 and New York City Response and Recovery Operations. December. Accessed November 2, 2012, at: <u>www.epa.gov/wtc/reports/residential_dust_cleanup_final_report.pdf</u>

US EPA. 2008a. *Child-Specific Exposure Factors Handbook. Final Report.* EPA/600/R-06/096F. Accessed November 1, 2012, at: http://cfpub.epa.gov/ncea/cfm/recordisplay.cfm?deid=199243

US EPA. 2008b. Guidance for the Sampling and Analysis of Lead in Indoor Residential Dust for Use in the Integrated Exposure Uptake Biokinetic (IEUBK) Model. Prepared by the Lead Committee of the Technical Review Workgroup for Metals and Asbestos. OSWER 9285.7–81. December. Accessed November 2, 2012, at: <u>http://epa.gov/superfund/lead/products/dust_sampling_guidance_final.pdf</u>

United States Department of Housing and Urban Development (US HUD). 2006 (revised). *Study of HUDs Risk Assessment Methodology in Three US Communities*. Accessed October 30, 2012, at: <u>www.nchh.org/Portals/0/Contents/Risk Assessment Final Report.pdf</u>

US HUD. 2012. *Guidelines for the Evaluation and Control of Lead-Based Paint Hazards in Housing* (Second Edition). Office of Healthy Homes and Lead Hazard Control. July. Accessed November 2, 2012, at: <u>http://portal.hud.gov/hudportal/HUD?src=/</u> program_offices/healthy_homes/lbp/hudguidelines

Van Holderbeke M, Cornelis C, Bierkens J, Torfs R. 2008. Review of the Soil Ingestion Pathway in Human Exposure Assessment. Study in Support of the BeNeKempen Project—Subproject on Harmonization of the Human Health Risk Assessment Methodology, Final Report. Mechelen, Belgium: VITO. 2008/IMS/R/0151.

Verein Deutscher Ingenieure (VDI, Association of German Engineers). 2001. German Protocol VDI 4300, Part 8. Measurement of Indoor Air Pollution: Sampling of House Dust. June. Withdrawn August 2012. [Revised December 2013 Version (Part 11) was released under the title of Measurement of indoor air pollution—Measurement strategies for determination of airborne particles in indoor environment— $PM_{2.5}$ fraction]

von Lindern I, Spalinger S, Petroysan V, von Braun M. 2003. Assessing remedial effectiveness through the blood lead:soil/dust lead relationship at the Bunker Hill Superfund Site in the Silver Valley of Idaho. *Science of the Total Environment* 303: 139–170.

Wilson R, Jones-Otazo H, Petrovic S, Mitchell I, Bonvalot Y, Wiliams D, Richardson GM. 2013. Revisiting dust and soil ingestion rates based on hand-to-mouth transfer. *Human and Ecological Risk Assessment* 19(1): 158–188.

World Trade Center Indoor Air Task Force Working Group. 2003. World Trade Center Indoor Environment Assessment: Selecting Contaminants of Potential Concern and Setting Health-Based Benchmarks. Prepared by the Contaminants of Potential Concern (COPC) Committee of the World Trade Center Indoor Air Task Force Working Group. May. Accessed November 1, 2012, at: <u>www.epa.gov/wtc/reports/contaminants_of_concern_benchmark_study.pdf</u>

Xue J, Zartarian V, Moya J, Freeman N, Beamer P, Black K, Tulve N, Shalat S. 2007. A meta-analysis of children's hand-to-mouth frequency data for estimated non-dietary ingestion exposure. *Risk Analysis*. 27(2): 411–420.

Yiin LM, Rhoads GG, Lioy PJ. 2000. Seasonal influences on childhood lead exposure. *Environmental Health Perspectives* 108(2): 177–182.

Yu C, Yiin L, Lioy P. 2006. The bioaccessibility of lead (Pb) from vacuumed house dust on carpets in urban residences. *Risk Analysis* 26(1): 125–134.

Zhang J, Smith KR, Ma Y, Ye S, Jiang F, Qi W, Liu P, Khalil MAK, Rasmussen RA, Thorneloe SA. 2000. Greenhouse gases and other pollutants from household stoves in China: a database for emission factors. *Atmospheric Environment* 34: 4537–4549.

APPENDIX I

CALCULATION EQUATIONS FOR DSC_{HH}

Equations for calculating human health effects-based dust screening concentrations (DSC_{HH}) were adapted from approaches identified by CCME (2006) and WTCWG (2003).

Equations to calculate DSC_{HH} are identified for a residential scenario, a commercial scenario (where people of all ages may be present) and a constrained use commercial scenario (where only adults are expected to spend appreciable amounts of time).

Effects-Based Dust Concentrations—Surface Area Concentrations, Non-Carcinogens

For non-carcinogens, DSC_{HH} expressed on a mass per surface area basis can be calculated as follows:

(Eq.1)
$$DSC_{HH} = \frac{THQ \times TDI \times BW}{[(SA_{fingers} \times FQ \times SE \times RAF_{oral} \times EFF) + (TC \times RAF_{derm} \times EFB)] \times EF}$$

Where:		
DSC _{HH}	=	human health effects-based dust screening concentrations (µg/m ²)
THQ	=	target hazard quotient (unitless)
TDI	=	tolerable daily intake (µg/kg-d)
BW	=	body weight (kg)
SA _{fingers}	=	surface area of fingers inserted into mouth (m ² ; from Eq. 2)
FQ	=	frequency of hand-to-mouth events (hr ⁻¹)
SE	=	saliva extraction factor (unitless)
RAF	=	relative oral absorption factor (unitless)
EFF	=	exposure factor for transfer to fingertips (hr/d; from Eq. 3)
TC	=	transfer coefficient (m ² /hr). The TC represents the rate of skin contact with the surface.
RAF	=	relative dermal absorption factor (unitless)
EFB	=	exposure factor for transfer to body (hr/d; from Eq. 4)
EF	=	exposure frequency (unitless)

The most sensitive receptors for surface area DSC_{HH} for non-carcinogens are toddler (residential), toddler (commercial) and adult (constrained use commercial).

To calculate the surface area of fingers inserted into mouth:

(Eq.2)
$$SA_{fingers} = \frac{SA_{hands}}{2} \times FSA_{fingers}$$

Where:

SA _{fingers}	=	surface area of fingers inserted into mouth (m ²)
SA	=	surface area of hands exposed to dust (both hands; m ²)
FSA _{fingers}	=	fractional surface area of fingers for mouthing activity (unitless)

To calculate the exposure factor for transfer to finger tips:

(Eq.3)
$$EFF = (HPD - TO - ST) \times (0.5 \times FTSS_{HSH} + 0.5 \times FTSS_{SSH})$$

Where:		
EFF	=	exposure factor for transfer to fingertips (hr/d)
HPD	=	hours spent in environment each day (hr/d)
ТО	=	time spent outdoors (hr/d)
ST	=	sleep time (hr/d)
FTSS _{HSH}	=	fraction of dust transferred from hard surfaces to hands (unitless)
FTSS	=	fraction of dust transferred from soft surfaces to hands (unitless)

To calculate the exposure factor for transfer to the body:

(Eq. 4)	$FFB = (HPD - TO - ST) \times$	$(0.5 \times \text{FTSS} + 0.5 \times \text{FTSS})$
(Eq.4)	$EFD = (IIID = IO = SI) \land$	$(0.3 \land 1.1 \Im_{HSB} + 0.3 \land 1.1 \Im_{SSB})$

Where:EFB=exposure factor for transfer to body (hr/d)HPD=hours spent in environment each day (hr/d)TO=time spent outdoors (hr/d)ST=sleep time (hr/d)FTSS
HSB=fraction of dust transferred from hard surfaces to body (unitless)FTSS
SSR=fraction of dust transferred from soft surfaces to body (unitless)

Effects-Based Dust Concentrations—Surface Area Concentrations, Carcinogens

For carcinogens, DSC_{HH} expressed on a mass per surface area basis can be calculated as follows:

(Eq.5)
$$DSC_{HH} = \frac{TILCR \times BW}{[(SA_{fingers} \times FQ \times SE \times RAF_{oral} \times EFF) + (TC \times RAF_{derm} \times EFB)] \times EF \times CSF}$$

Where:		
DSC	=	human health effects-based dust screening concentration (µg/m ²)
TILCR	=	target incremental lifetime cancer risk (unitless)
BW	=	body weight (kg)
SA _{fingers}	=	surface area of fingers inserted into mouth (m ² ; from Eq. 2)
FQ	=	frequency of hand-to-mouth events (hr ⁻¹)
SE	=	saliva extraction factor (unitless)
RAF	=	relative oral absorption factor (unitless)
EFF	=	exposure factor for transfer to fingertips (hr/d; from Eq. 3)
TC	=	transfer coefficient (m ² /hr). The TC represents the rate of skin contact with the surface.
RAF	=	relative dermal absorption factor (unitless)
EFB	=	exposure factor for transfer to body (hr/d; from Eq. 4)
EF	=	exposure frequency (unitless)
CSF	=	cancer slope factor ([µg/kg-d] ⁻¹)

Effects-Based Dust Concentrations—Bulk Dust Concentrations, Non-Carcinogens

For non-carcinogens, DSC_{HH} expressed on a mass per mass (bulk dust) basis can be calculated as follows:

(Eq.6)
$$DSC_{HH} = \frac{THQ \times TDI \times BW}{(DIG \times RAF_{oral} + [(SAF_{hands} \times SA_{hands}) + (SAF_{other} \times SA_{other})] \times RAF_{derm}) \times EF}$$

Where:		
DSC _{HH}	=	human health effects-based dust screening concentration (μ g/g)
THQ	=	target hazard quotient (unitless)
TDI	=	tolerable daily intake (µg/kg-d)
BW	=	body weight (kg)
DIG	=	dust ingestion rate (g/d; see Eq.8)
RAF	=	relative oral absorption factor (unitless)
SAF	=	skin adherence factor for dust on hands (g/m ² -d)
SA _{hands}	=	surface area of hands exposed to dust (both hands; m ²)
SAF	=	skin adherence factor for dust on other body parts (g/m ² -d)
SA _{other}	=	surface area of other body parts exposed to dust (m ²)
RAF	=	relative dermal absorption factor (unitless)
EF	=	exposure frequency (unitless)

The most sensitive receptors for bulk dust DSC_{HH} for non-carcinogens are toddler (residential), toddler (commercial) and adult (constrained use commercial).

$$(Eq.7) \qquad DIG = \frac{(DSL_{hard} \times \frac{SA_{hands}}{2} \times FSA_{fingers} \times FTSS_{HSH} \times FQ \times SE \times ET_{hard})}{1000} + \frac{(DSL_{soft} \times \frac{SA_{hands}}{2} \times FSA_{fingers} \times FTSS_{SSH} \times FQ \times SE \times ET_{soft})}{1000}$$

=	dust ingestion rate (g/d) (see Eq. 8)
=	dust surface loading on horizontal hard surfaces (mg/cm ²)
=	surface area of hands exposed to dust (both hands; cm ²)
=	fractional surface area of fingers for mouthing activity (unitless)
=	fraction of dust transferred from hard surfaces to hands (unitless)
=	frequency of hand-to-mouth events (hr ⁻¹)
=	saliva extraction factor (unitless)
=	exposure time for contact with horizontal hard surfaces (hr/d; see Eq. 8)
=	dust surface loading on horizontal soft surfaces (mg/cm ²)
=	fraction of dust transferred from soft surfaces to hands (unitless)
=	exposure time for contact with horizontal soft surfaces (hr/d; see Eq. 9)
=	conversion factor for mg/d to g/d

To calculate exposure time for contact with horizontal hard surfaces:

$$(Eq.8)^{1,2} ET_{hard} = (HPD - TO - ST) \times 0.5$$

Where:		
ET _{hard}	=	exposure time for contact with horizontal hard surfaces (hr/d)
HPD	=	hours spent in environment each day (hr/d)
ТО	=	time spent outdoors (hr/d)
ST	=	sleep time (hr/d)
0.5	=	factor to account for 50% of time exposed to hard surfaces and 50% of time exposed to soft surfaces (McDonald <i>et al.</i> , 2011)

To calculate exposure time for contact with horizontal soft surfaces:

$$(Eq.9)^{2,3}$$
 $ET_{soft} = (HPD - TO - ST) \times 0.5$

Where:		
ET _{soft}	=	exposure time for contact with horizontal soft surfaces (hr/d)
HPD	=	hours spent in environment each day (hr/d)
ТО	=	time spent outdoors (hr/d)
ST	=	sleep time (hr/d)
0.5	=	factor to account for 50% of time exposed to hard surfaces and 50% of time exposed to soft surfaces (McDonald et al., 2011)

¹ Equation 8 is not relevant to infants

² In equations 8 and 9, the variables TO and ST are not applicable to a commercial scenario.

³ Multiplication by 0.5 is not required for the infant age group (i.e., all time spent by the infant is assumed to be with soft surfaces)

Effects-Based Dust Concentrations—Bulk Dust Concentrations, Carcinogens

For carcinogens, DSC_{HH} expressed on a mass per mass bulk dust basis can be calculated as follows:

(Eq.10)		DSC =
(14.10)		$(\text{DIG} \times \text{RAF}_{\text{oral}} + [(\text{SAF}_{\text{hands}} \times \text{SA}_{\text{hands}}) + (\text{SAF}_{\text{other}} \times \text{SA}_{\text{other}})] \times \text{RAF}_{\text{derm}}) \times \text{EF} \times \text{CSF}$
Where:		
DSC _{HH}	=	human health effects-based dust screening concentration (μg/g)
TILCR	=	target incremental lifetime cancer risk (unitless)
BW	=	body weight (kg)
DIG	=	dust ingestion rate (g/d; see Eq. 8)
RAF	=	relative oral absorption factor (unitless)
SAF	=	skin adherence factor for dust on hands (g/m ² -d)
SA	=	surface area of hands exposed to dust (both hands; m ²)
SAF _{other}	=	skin adherence factor for dust on other body parts (g/m ² -d)
SA _{other}	=	surface area of other body parts exposed to dust (m ²)
RAF	=	relative dermal absorption factor (unitless)
EF	=	exposure frequency (unitless)
CSF	=	cancer slope factor ([µg/kg-d] ⁻¹)

APPENDIX II

EQUATIONS FOR EXPOSURE ASSESSMENT OF DUST IN HHRAs

Within an HHRA, the equations used to estimate exposure to COPCs in dust are slightly different from those used to calculate DSC_{HH} which can be used to screen COPCs in dust. Equations are presented below for assessment of exposure to indoor dust in an HHRA. These exposure calculations would then be integrated with a multi-media exposure assessment.

Estimation of Oral Exposure on an Area-Weighted Basis

$$(Eq.1) \qquad EID_{total} = \frac{DCA_{hard} \times ET_{hard} \times FTSS_{HSH} \times SA_{fingers} \times FQ \times SE \times RAF_{oral} \times EF}{BW} \\ + \frac{DCA_{soft} \times ET_{soft} \times FTSS_{SSH} \times SA_{fingers} \times FQ \times SE \times RAF_{oral} \times EF}{BW}$$

Where:		
EID	=	exposure from dust ingestion from hard and soft surfaces (μ g/kg-d)
DCA	=	dust concentration from hard surfaces on an area-weighted basis (µg/m ²)
ET	=	exposure time for contact with horizontal hard surfaces (hr/d; from Eq. 2)
FTSS _{HSH}	=	fraction of dust transferred from hard surfaces to hands (unitless)
SA _{fingers}	=	surface area of fingers inserted into mouth (m ² ; from Eq. 3)
FQ	=	frequency of hand-to-mouth events (hr ⁻¹)
SE	=	saliva extraction factor (unitless)
RAF _{oral}	=	relative oral absorption factor (unitless)
EF	=	exposure frequency (unitless)
BW	=	body weight (kg)
DCA _{soft}	=	dust concentration from soft surfaces on an area-weighted basis (μ g/m ²)
ET	=	exposure time for contact with horizontal soft surfaces (hr/d; from Eq. 4)
FTSS	=	fraction of dust transferred from soft surfaces to hands (unitless)

To calculate exposure time for contact with horizontal hard surfaces:

(Eq.2)
$$ET_{hard} = (HPD - TO - ST) \times 0.5$$

Where:		
ET _{hard}	=	exposure time for contact with horizontal hard surfaces (hr/d)
HPD	=	hours spent in environment each day (hr/d)
ТО	=	time spent outdoors (hr/d)
ST	=	sleep time (hr/d)
0.5	=	factor to account for 50% of time exposed to hard surfaces and 50% of time exposed to soft surfaces (McDonald et al., 2011).

To calculate the surface area of fingers inserted into mouth:

(Eq.3)
$$SA_{fingers} = \frac{SA_{hands}}{2} \times FSA_{fingers}$$

Where:		
SA _{fingers}	=	surface area of fingers inserted into mouth (m ²)
SA	=	surface area of hands exposed to dust (both hands; m ²)
FSA _{fingers}	=	fractional surface area of fingers for mouthing activity (unitless)

To calculate exposure time for contact with horizontal soft surfaces:

(Eq.4)
$$ET_{soft} = (HPD - TO - ST) \times 0.5$$

Where:		
ET _{soft}	=	exposure time for contact with horizontal soft surfaces (hr/d)
HPD	=	hours spent in environment each day (hr/d)
ТО	=	time spent outdoors (hr/d)
ST	=	sleep time (hr/d)
0.5	=	factor to account for 50% of time exposed to hard surfaces and 50% of time exposed to soft surfaces (McDonald et al., 2011).

Estimation of Dermal Exposure on an Area-Weighted Basis

$$(Eq.5) \qquad EDD_{total} = \frac{DCA_{hard} \times TC \times ET_{hard} \times FTSS_{HSH} \times RAF_{dermal} \times EF}{BW} + \frac{DCA_{hard} \times TC \times ET_{soft} \times FTSS_{SSH} \times RAF_{dermal} \times EF}{BW}$$

Where:		
EDD _{total}	=	exposure from dermal contact with indoor dust from hard and soft surfaces (μ g/kg-d)
DCA _{hard}	=	dust concentration from hard surfaces on an area-weighted basis (μ g/m ²)
TC	=	transfer coefficient (m ² /hr)
ET _{hard}	=	exposure time for contact with horizontal hard surfaces (hr/d; from Eq. 2)
FTSS _{HSH}	=	fraction of dust transferred from hard surfaces to hands (unitless)
RAF	=	relative dermal absorption factor (unitless)
EF	=	exposure frequency (unitless)
BW	=	body weight (kg)
DCA _{soft}	=	dust concentration from soft surfaces on an area-weighted basis (µg/m ²)
ET	=	exposure time for contact with horizontal soft surfaces (hr/d; from Eq. 4)
FTSS _{SSH}	=	fraction of dust transferred from soft surfaces to hands (unitless)

Estimation of Total Exposure on an Area-Weighted Basis

(Eq.6)		$EAD_{total} = EID_{total} + EDD_{total}$
Where:		
EAD _{total}	=	exposure to dust from ingestion and dermal pathways (μ g/kg-d)
EID _{total}	=	exposure from dust ingestion from hard and soft surfaces (µg/kg-d)
EDD	=	exposure from dermal contact with indoor dust from hard and soft surfaces (μ g/kg-d)

Estimation of Oral Exposure from Bulk Dust Concentrations

(Eq.7)
$$EID = \frac{BDC \times DIG \times RAF_{oral} \times EF}{BW}$$

Where:		
EID	=	exposure from dust ingestion from horizontal surfaces (µg/kg-d)
BDC	=	bulk dust concentration on horizontal surfaces (µg/g)
DIG	=	dust ingestion rate (g/d; see Eq. 8)
RAF	=	relative oral absorption factor (unitless)
EF	=	exposure frequency (unitless)
BW	=	body weight (kg)

To calculate the dust ingestion rate:

$$(Eq.8) \qquad DIG = \frac{(DSL_{hard} \times \frac{SA_{hands}}{2} \times FSA_{fingers} \times FTSS_{HSH} \times FQ \times SE \times ET_{hard})}{1000} \\ + \frac{(DSL_{soft} \times \frac{SA_{hands}}{2} \times FSA_{fingers} \times FTSS_{SSH} \times FQ \times SE \times ET_{soft})}{1000}$$

Where: DIG = dust ingestion rate (g/d) $\mathsf{DSL}_{\mathsf{hard}}$ = dust surface loading on horizontal hard surfaces (mg/cm²) $\mathsf{SA}_{\mathrm{hands}}$ = surface area of hands exposed to dust (both hands; cm²) $\mathsf{FSA}_{\mathrm{fingers}}$ = fractional surface area of fingers for mouthing activity (unitless) FTSS = fraction of dust transferred from hard surfaces to hands (unitless) FQ = frequency of hand-to-mouth events (hr^{-1}) SE = saliva extraction factor (unitless) $\mathsf{ET}_{\mathsf{hard}}$ = exposure time for contact with horizontal hard surfaces (hr/d; see Eq. 2) DSL_{soft} = dust surface loading on horizontal soft surfaces (mg/cm²) FTSS_{SSH} = fraction of dust transferred from soft surfaces to hands (unitless) $\mathsf{ET}_{\mathsf{soft}}$ = exposure time for contact with horizontal soft surfaces (hr/d; see Eq. 4) 1000 = conversion factor for mg/d to g/d

Estimation of Dermal Exposure from Bulk Dust Concentrations

(Eq.9)
$$EDD = \frac{BDC \times ([SAF_{hands} \times SA_{hands}] + [SAF_{other} \times SA_{other}]) \times RAF_{dermal} \times EF}{BW}$$

Where:		
EDD	=	exposure from dermal contact with indoor dust from horizontal surfaces (µg/kg-d)
BDC	=	bulk dust concentration on horizontal surfaces (µg/g)
SAF	=	skin adherence factor for dust on hands (g/m ² -d)
SA _{hands}	=	surface area of hands exposed to dust (both hands; m ²)
SAF	=	skin adherence factor for dust on other body parts (g/m ² -d)
SA _{other}	=	surface area of other body parts exposed to dust (m ²)
RAF	=	relative dermal absorption factor (unitless)
EF	=	exposure frequency (unitless)
BW	=	body weight (kg)

Estimation of Total Exposure from Bulk Dust Concentrations

(Eq.10)
$$EAD_{total} = EID_{total} + EDD_{total}$$

Where:

EAD _{total}	=	exposure to dust from ingestion and dermal pathways (µg/kg-d)
EID	=	exposure from dust ingestion from horizontal surfaces (µg/kg-d)
EDD _{total}	=	exposure from dermal contact with indoor dust from horizontal surfaces ($\mu\text{g/kg-d})$